The Minister for Planning

Goodman Property Services (Aust) Pty Limited

The Trust Company Limited as custodian for the Carter Street Trust

The Trust Company Limited as custodian for the Hill Road Trust

The Trust Company Limited as custodian for the IBC Trust

Tallina Pty Ltd as trustee for the Carter Street Trust

Tallina Pty Ltd as trustee for the Hill Road Trust

Tallina Pty Ltd as trustee for the IBC Trust

Planning Agreement

Section 93F of the Environmental Planning and Assessment Act, 1979 (NSW)

Contents

| 1 | Planning agreement under the Act | 2 | | |
|----|--|----------------------------------|--|--|
| 2 | Application of this document | | | |
| 3 | Operation and commencement of this document | | | |
| 4 | Definitions | 2 | | |
| 5 | Development Contributions to be made under this document | 9 | | |
| | 5.1 Developer and Landowners to provide Development Contribution5.2 Acknowledgement | itions 9 | | |
| | 5.3 Special Infrastructure Contributions | 9 | | |
| 6 | Interest for late payment | 10 | | |
| 7 | Application of sections 94, 94A and 94EF of the Act to the Development 10 | | | |
| 8 | Registration | | | |
| | 8.1 Registration of this document 8.2 Evidence of Registration 8.3 Right to lodge caveat 8.4 Release and discharge of document 8.5 Landowner's interest in Land | 10 12 12 14 14 | | |
| 9 | Dispute Resolution | 15 | | |
| | 9.1 Reference to dispute 9.2 Notice of dispute 9.3 Representatives of Parties to meet 9.4 Mediation 9.5 Court proceedings 9.6 Not use information 9.7 No prejudice | 15 15 15 15 15 15 | | |
| 10 | Enforcement | 16 | | |
| | 10.1 Developer and Landowners to provide security10.2 Enforcement by any Party | 16 16 | | |
| 11 | Notices | 16 | | |
| | 11.1 Delivery11.2 Change of details11.3 Giving of notice11.4 Delivery outside of business hours | 16 18 18 18 | | |
| 12 | Approvals and consent | 18 | | |
| 13 | Assignment and dealings | 18 | | |
| | 13.1 Restriction on assignment and novation13.2 Restriction on Dealings | 18 19 | | |

| 14 | Capa | city | 22 |
|----|-------|--|----|
| 15 | Repo | rting requirement | 22 |
| 16 | Limit | 23 | |
| | 16.1 | Definitions | 23 |
| | 16.2 | Limitation of the First Landowner's Liability | 24 |
| | 16.3 | Limitation of Trustee's Liability | 25 |
| 17 | Limit | ation of liability for the Hill Road Trust | 26 |
| | 17.1 | Definitions | 26 |
| | 17.2 | Limitation of the Second Landowner's Liability | 27 |
| | 17.3 | Limitation of Trustee's Liability | 28 |
| 18 | Limit | ation of liability for the IBC Trust | 29 |
| | 18.1 | Definitions | 29 |
| | 18.2 | Limitation of the Third Landowner's Liability | 30 |
| | | Limitation of Trustee's Liability | 31 |
| 19 | Trus | tee's representations and warranties | 32 |
| | 19.1 | Definitions | 32 |
| | 19.2 | The Trustee's representations and warranties | 33 |
| | 19.3 | Trustee's additional obligations | 35 |
| | 19.4 | The Landowners' representations and warranties | 36 |
| | 19.5 | The Landowners' and Trustees' additional obligations | 36 |
| | 19.6 | Non Merger | 37 |
| 20 | GST | | 37 |
| | 20.1 | Construction | 37 |
| | 20.2 | Consideration GST exclusive | 37 |
| | 20.3 | Payment of GST | 37 |
| | 20.4 | Timing of GST payment | 38 |
| | 20.5 | Tax invoice | 38 |
| | 20.6 | Adjustment event | 38 |
| | 20.7 | Reimbursements | 38 |
| | 20.8 | Calculations based on other amounts | 38 |
| | 20.9 | No merger | 38 |
| 21 | Gene | ral provisions | 39 |
| | 21.1 | Costs, expenses and stamp duty | 39 |
| | 21.2 | Entire agreement | 39 |
| | 21.3 | Variation | 39 |
| | 21.4 | Further acts | 40 |
| | 21.5 | Governing law and jurisdiction | 40 |
| | 21.6 | Joint and individual liability and benefits | 40 |
| | 21.7 | No fetter | 40 |
| | 21.8 | Representations and warranties | 40 |
| | 21.9 | Severability | 40 |
| | 21.10 | Waiver | 41 |

Corrs Chambers Westgarth

| 21.11 Effect of Schedules | 41 |
|--|----|
| 21.12 Relationship of parties | 41 |
| 21.13 Further steps | 41 |
| 21.14 Counterparts | 41 |
| 21.15 Rights cumulative | 41 |
| 21.16 Liability | 41 |
| 21.17 Interpretation | 41 |
| 21.18 Headings | 42 |
| 21.19 Effect of execution | 42 |
| Schedule 1 - Section 93F Requirements | 43 |
| Schedule 2 - Land | 45 |
| Schedule 3 - Development Contributions | 47 |
| Schedule 4 - Security | 54 |
| Schedule 5 - Explanatory Note | 57 |
| Execution | 61 |
| Annexure A – Novation Deed/Assignment Deed | 65 |
| Annexure B - Plan | 66 |
| Annexure C - Education Land | 67 |
| Annexure D - Form of Bank Guarantee | 68 |
| Annexure E - Form of Caveat | 69 |

Date

Parties

The Minister for Planning ABN 38 755 709 681 of Level 15, 52 Martin Place, Sydney NSW 2000 (**Minister**)

Goodman Property Services (Aust) Pty Limited ABN 40 088 981 793 of Level 17, 60 Castlereagh Street, Sydney NSW 2000 (**Developer**)

The Trust Company Limited ABN 59 004 027 749 of Level 12, 123 Pitt Street Sydney NSW 2000 as custodian for the Carter Street Trust ABN 12 909 843 665 (**First Landowner**)

The Trust Company Limited ABN 59 004 027 749 of Level 12, 123 Pitt Street Sydney NSW 2000 as custodian for the Hill Road Trust ABN 54 244 152 320 (**Second Landowner**)

The Trust Company Limited ABN 59 004 027 749 of Level 12, 123 Pitt Street Sydney NSW 2000 as custodian for the IBC Trust ABN 30 476 081 229 (**Third Landowner**)

Tallina Pty Ltd ACN 090 716 895 of Level 17, 60 Castlereagh Street, Sydney NSW 2000 as trustee for the Carter Street Trust ABN 12 909 843 665

Tallina Pty Ltd ACN 090 716 895 of Level 17, 60 Castlereagh Street, Sydney NSW 2000 as trustee for the Hill Road Trust ABN ABN 54 244 152 320

Tallina Pty Ltd ACN 090 716 895 of Level 17, 60 Castlereagh Street, Sydney NSW 2000 as trustee for the IBC Trust ABN 30 476 081 229

Background

- A The First Landowner owns part of the Land as custodian for the Carter Street Trust.
- B The Second Landowner owns part of the Land as custodian for the Hill Road Trust.
- C The Third Landowner owns part of the Land as custodian for the IBC Trust.
- D The Land is located in the Carter Street Priority Precinct.
- E The Developer has requested that the Department progress the Planning Proposal seeking to rezone the Land.
- F The Landowners and the Developer have offered to enter into this document to make Development Contributions towards the Public Purpose following the operation of this Agreement.

Operative provisions

1 Planning agreement under the Act

The Parties agree that this document is a planning agreement governed by subdivision 2 of Division 6 of Part 4 of the Act.

2 Application of this document

This document applies to:

- (a) the Land; and
- (b) the Development.

3 Operation and commencement of this document

This document will commence from the later of:

- (a) commencement of the Proposed SEPP within the meaning of section 34(5) of the Act; and
- (b) this document being entered into in accordance with clause 25C(1) of the Regulation.

4 Definitions

In this document the following terms have the following meanings:

Act The Environmental Planning and Assessment Act

1979 (NSW).

ALEP The Auburn Local Environmental Plan 2010, as

amended from time to time.

Allowable GFA The maximum GFA permitted to be developed under

the ALEP on the relevant land.

Authority Any federal, State or local government or semi-

governmental, statutory, judicial or public person,

instrumentality or department.

Bank Guarantee An irrevocable and unconditional undertaking which is

substantially in the form of the undertaking attached at **Annexure D** and issued by an Australian bank which is an eligible financial institution for the purposes of Treasury Circular NSW TC14/01 dated 24 January 2014 as amended, supplemented or substituted from

time to time.

Business Day

A day which is not a Saturday, Sunday or bank or public holiday in Sydney.

Carter Street Priority Precinct

The 52ha precinct in the Auburn local government area that adjoins Sydney Olympic Park, as described in the Planning Proposal and the Planning Report.

Chemical Substance

Without limitation:

- (a) any chemical substance or mixture, including any pollutant, contaminant, chemical, raw material or intermediate whether solid, liquid, gaseous or otherwise;
- (b) industrial solid, toxic or hazardous substance, material or waste:
- (c) petroleum or any fraction thereof;
- (d) asbestos or asbestos-containing material (including asbestos in structures);
- (e) persistent organic pollutants and chlorinated hydrocarbons;
- (f) heavy metals, such as mercury, cadmium and lead;
- (g) radionuclides;
- (h) volatile organic compounds;
- solvents; and
- (j) pesticides,

and includes by-products and derivatives of the above and all substances, materials or wastes that are identified in or regulated under any Environmental Law.

Claim

Any allegation, debt, cause of action, liability, claim, proceeding, suit or demand of any nature howsoever arising and whether present or future, fixed or unascertained, actual or contingent whether at law, in equity, under statute or otherwise, including (without limitation), any claim for compensation arising under or pursuant to the Just Terms Act.

Complying Development Certificate

Has the same meaning as in the Act.

Construction Certificate

Has the same meaning as in the Act.

Contamination

Has the meaning given to that term in the Contaminated Land Management Act 1997 (NSW). Contaminate and Contaminated have corresponding

meanings.

Contribution Amount

The amount of the monetary contribution to be paid by the Developer or relevant Landowner as described in Schedule 3.

Contribution Area Contribution Area 1, Contribution Area 2, Contribution

Area 3, Contribution Area 4 or Contribution Area 5, as

each of those lots is identified in the Plan.

Contribution Areas Contribution Area 1, Contribution Area 2, Contribution Area 3. Contribution Area 4 and Contribution Area 5.

as each of those lots is identified in the Plan.

Contribution Area 1 That part of the Land described at Item 1 of **Schedule**

2 and identified as 'Contribution Area 1' on the Plan.

Contribution Area 2 That part of the Land described at Item 2 of **Schedule**

2 and identified as 'Contribution Area 2' on the Plan.

Contribution Area 3 That part of the Land described at Item 3 of **Schedule**

2 and identified as 'Contribution Area 3' on the Plan.

Contribution Area 4 That part of the Land described at Item 4 of **Schedule**

2 and identified as 'Contribution Area 4' on the Plan.

That part of the Land described at Item 5 of **Schedule** Contribution Area 5

2 and identified as 'Contribution Area 5' on the Plan.

The Sydney Consumer Price Index (All Groups) published by the Australian Bureau of Statistics or, if that index no longer exists, any similar index which the

Minister may determine in its sole discretion.

Dealing Means:

CPI

in relation to the Education Land, without (a) limitation, selling, transferring, assigning, charging, encumbering or otherwise dealing with

the Education Land; and

in relation to any part of the Land or any Contribution Area other than the Education Land, selling, transferring or assigning the relevant part of the Land or relevant Contribution

Area.

Department The Department of Planning and Environment, formerly the Department of Planning & Infrastructure,

or any other department replacing it from time to time.

Developer-Related Entity

The following parties:

- Tallina Pty Ltd ABN 68 090 716 895; (a)
- (b) Goodman Limited ABN 69 000 123 071;

- (c) Goodman Funds Management Limited ABN 067 796 641;
- (d) Goodman Funds Management Australia Limited ABN 13 113 249 595; and
- (e) any other Related Body Corporate of the Developer.

Development

Subdivision and development of the Land for residential, retail and commercial purposes, and any further subdivision of the Contribution Areas once they are created as separate parcels, but excluding:

- (a) the subdivision of the Land for the purpose of creating parcels comprising the whole of each Contribution Area as identified on the Plan and any roads which are required to be created for the purpose of creating each Contribution Area; and
- (b) the subdivision of Contribution Area 3 for the purpose of creating a separate title for the Education Land as identified on the Plan; and
- (c) Remediation of the Education Land in accordance with the terms of this document.

Development Contribution

To dedicate land free of cost, pay a monetary contribution, or provide any other material public benefit, or any combination of them, to be used for or applied towards a Public Purpose.

Education Land

The minimum 2 hectare parcel of land shown on the plan attached at **Annexure C** intended to be used for the purposes of a primary school and for public sports and playing fields. [Note, a survey showing the boundaries of the Education Land will be added to Annexure C before execution by the parties]

EnergyAustralia Land

Lot 200 in DP1160458, being part of the Land, which, as at the date of commencement of this document, is owned by EnergyAustralia.

Environment

Has the meaning given to that term in the Contaminated Land Management Act 1997 (NSW).

Environmental Laws

Any law (including statute, policies and regulations enforceable by statute) concerning the Environment, including, without limitation, laws concerning:

- (a) emissions of Chemical Substances into the atmosphere, waters and land;
- (b) pollution or Contamination of the atmosphere,

waters and land; and

- (c) production, use, handling, storage, transportation and disposal of:
 - (i) waste;
 - (ii) Hazardous Materials;
 - (iii) dangerous goods; and
 - (iv) Chemical Substances.

Explanatory Note

The Explanatory Note attached at **Schedule 5**.

First Landowner

The Trust Company Limited as custodian for the

Carter Street Trust.

GFA

Gross floor area, as that term is defined in the

Standard Instrument LEP.

Goodman

Goodman Property Services (Aust) Pty Limited ABN

40 088 981 793, the Developer.

GST

Has the same meaning as in the GST Law.

GST Law

Has the meaning given to that term in *A New Tax System (Goods and Services Tax) Act 1999* (Cth) and any other Act or regulation relating to the imposition or administration of the GST.

Hazardous Material

Any substance, gas, liquid, chemical, mineral or other physical or biological matter that is:

- (a) or may become toxic, flammable or inflammable;
- (b) otherwise dangerous, harmful to the Environment or any life form or which may cause pollution, Contamination or any hazard or increase in toxicity in the Environment or may leak or discharge or otherwise cause damage to any person, property or the Environment; or
- (c) a material or compound controlled, prohibited or regulated from time to time by any Environmental Law.

Just Terms Act

The Land Acquisition (Just Terms Compensation) Act 1991 (NSW).

Land

The whole or any part of the land comprising Contribution Area 1, Contribution Area 2, Contribution Area 3, Contribution Area 4 and Contribution Area 5 as described in **Schedule 2**, but excluding the Sydney Water Lot.

Landowner

The First Landowner or the Second Landowner or the Third Landowner as applicable to the relevant part of

the Land, as set out in **Schedule 2**. **Landowners** means the First Landowner, the Second Landowner and the Third Landowner.

Loss

Any loss (including loss of profit and loss of expected profit), claim, action, liability, damage, cost, charge, expense, outgoing, payment, diminution in value or deficiency of any kind or character which the Minister pays, suffers or incurs or is liable for, including:

- (a) liabilities on account of Tax;
- (b) interest and other amounts payable to third parties; and
- (c) legal (on a full indemnity basis) and other expenses incurred in connection with investigating or defending any claim or action, whether or not resulting in any liability, and all amounts paid in settlement of any claim or action.

LPI

NSW Land and Property Information or any other government agency replacing it.

Novation Deed

The deed attached at Annexure A.

Party

A party to this document, including their successors and assigns, but only to the extent that assignment is permitted under **clause 13**.

Permitted Lease

A new lease over any part of the Land which includes a term requiring the lessee to consent to the Registration of this document on the title to the relevant part of the Land subject to the lease.

Plan

The plan of the Land and the Contribution Areas, attached at **Annexure B**.

Planning Agreement

Has the same meaning as in the Act.

Planning Application A development application in respect of any part of the Land made under Part 4 of the Act.

Planning Approval

The determination by approval of a Planning Application.

Planning Proposal

The Carter Street Lidcombe Urban Activation Precinct Proposal prepared by the Department exhibited in March/April 2014 and dated February 2014 for the rezoning of the Carter Street Priority Precinct by means of the Proposed SEPP and associated

development control plan.

Planning Report

The Carter Street Lidcombe Urban Activation Precinct Planning Report prepared by the Department and

dated February 2014.

Proposed SEPP Any State Environmental Planning Policy to amend

the ALEP that is made generally in accordance with

the Planning Proposal.

Public Purpose Any purpose that benefits the public or a section of the

public, including those specified in section 93F(2) of

the Act.

Register Means the Torrens title register maintained under the

Real Property Act 1900 (NSW). Registration has the

corresponding meaning.

Regulation The Environmental Planning and Assessment

Regulation 2000 (NSW).

Related Body Corporate

Has the meaning given to that term in the

Corporations Act 2001 (Cth).

Remediation Of Contaminated land includes:

(a) the preparation of a long term management plan

for the land;

(b) investigating, treating, removing, disposing, destroying, reducing, mitigating, dispersing,

remediating or containing the Contamination of

the land; and/or

(c) eliminating or reducing any hazard arising from the Contamination of the land (including by

preventing the entry or persons or animals onto

the land).

Remediate has a corresponding meaning.

Satisfactory Arrangements Certificate A certificate issued by the Secretary confirming that the satisfactory arrangements provisions included in the Proposed SEPP pursuant to section 93D of the

Act have been satisfied.

Second Landowner The Trust Company Limited as custodian for the Hill

Road Trust.

Secretary The Secretary of the Department of Planning and

Environment, from time to time

SIC Determination A determination by the Minister of a Special

Infrastructure Contribution for a special contributions

area which includes the Land.

Site Audit Statement A site audit statement prepared by a Site Auditor in

accordance with Part 4 of the Contaminated Land

Management Act 1997.

Site Auditor A suitably qualified site auditor duly qualified under the

Contaminated Land Management Act 1997 (NSW).

Special A contribution determined in accordance with section Infrastructure 94EE of the Act which applies to development on any

Contribution part of the Land.

Standard Instrument The Standard Instrument – Principal Local

LEP Environmental Plan.

Subdivision Has the same meaning as in the Act. **Certificate**

Sydney Water Lot Lot 1 in DP 234801.

Tax A tax, duty (including stamp duty and any other

transaction duty), levy, impost, charge, fee (including a registration fee) together with all interest, penalties,

fines and costs concerning them.

Third Landowner The Trust Company Limited as custodian for the IBC

Trust.

5 Development Contributions to be made under this document

5.1 Developer and Landowners to provide Development Contributions

The Developer and the Landowners agree to provide to the Minister or the Minister's nominee the Development Contributions in accordance with the provisions of **Schedule 3**.

5.2 Acknowledgement

The Developer and the Landowners acknowledge and agree that:

- (a) the Minister must comply with section 93E of the Act, but the Minister:
 - does not make any representation or warranty that the Development Contributions must be used for a particular purpose; and
 - (ii) has no obligation to repay the Development Contributions; and
- (b) in circumstances where the Development Contributions are transferred to any Authority, the Minister has not made any representation or warranty that the Development Contributions will or must be used for a particular purpose by that Authority.

5.3 Special Infrastructure Contributions

(a) The Minister agrees that the Development Contributions in this document, to be made by the Developer and the Landowners, wholly fulfil each of the Developer's and the Landowners' obligations to make

- contributions towards State infrastructure arising from the Development of the Land.
- (b) The Parties further agree that if the Minister makes a SIC Determination after the commencement of this document that applies to the whole or any part of the Land, the Landowners will have no obligation to pay any Special Infrastructure Contributions as a consequence of that SIC Determination, nor will they be entitled to any credits or offsets as a consequence of that SIC Determination.
- (c) Nothing about this clause 5.3 restricts, offsets, gives rise to credits or otherwise reduces any local infrastructure contributions that may be made or be required to be made in conjunction with the Development. However, the Development Contributions under this document may be taken into consideration in determining the local contributions to be made in conjunction with the Development.

6 Interest for late payment

- (a) If the Developer and the Landowners fail to pay a Contribution Amount due to the Minister on the due date for payment, the Developer and the Landowners will be jointly liable to pay the Minister interest on the Contribution Amount, in addition to the Contribution Amount, at a rate of 2% above the loan reference rate charged by the Commonwealth Bank of Australia from time to time.
- (b) Interest will be payable on the daily balance of amounts due from the due date for payment of those amounts until all outstanding amounts (including interest on those amounts) have been paid to the Minister.

Application of sections 94, 94A and 94EF of the Act to the Development

The application of sections 94, 94A and 94EF of the Act are excluded to the extent stated in **Schedule 1**.

8 Registration

8.1 Registration of this document

- (a) The Developer and the Landowners agree that they will procure the Registration of this document on the relevant folios of the Register for the Land that is owned by each Landowner in accordance with section 93H of the Act and the terms of this document.
- (b) The Developer and the Landowners will:
 - (i) at the Developer's and the Landowners' expense; and

subject to clause 8.1(d) and clause 8.1(e), within 90 Business
 Days of receiving a copy of this document executed by the Minister,

take all practical steps and otherwise do all things necessary to procure, in relation to the Land, with the exception of the EnergyAustralia Land:

- (iii) the consent of each person who:
 - (A) has an estate or interest in the Land registered on the Register; and
 - (B) is seized or possessed of an estate or interest in the Land;
- (iv) the execution of any documents;
- (v) the production of the relevant certificates of title; and
- (vi) the lodgement of this document on the relevant folios of the Register relating to the Land.
- (c) The Developer and the Landowners must use all reasonable endeavours to procure Registration of this document on the relevant folios of the Register relating to the Land as soon as possible following lodgement.
- (d) Where the Developer and the Landowners are unable to procure Registration in relation to any part of the Land within 90 Business Days of receiving a copy of this document executed by the Minister, because a consent required from a third party has not been given, the relevant Party must:
 - (i) continue to use all reasonable endeavours to procure Registration of this document until it is Registered on all of the relevant folios of the Register relating to the Land;
 - (ii) in the event that the document has not been lodged over a particular Lot comprising part of the Land pending expiration of termination of a lease, lodge the document for Registration over any such Lot within 5 Business Days of the expiration or termination of that lease over that Lot; and
 - (iii) provide a written report to the Minister:
 - (A) on the date that is 90 Business Days after the Developer and the Landowners receive a copy of this document executed by the Minister; and
 - (B) thereafter monthly, on the first day of each month,
 - as to the steps which have been taken by the Developer and the Landowners to secure any required consents and to procure Registration of this document on title in respect of that Land.
- (e) With respect to the EnergyAustralia Land, the First Landowner will, at the Developer or the First Landowner's expense, promptly take all practical steps and otherwise do all things necessary to procure the matters in clauses 8.1(b)(iii) to (vi) within 10 Business Days of the completion of

- the purchase of the EnergyAustralia Land by the First Landowner or any Developer-Related Entity.
- (f) The Developer and the Landowners agree that they will not enter into any new lease over any part of the Land unless that lease is a Permitted Lease. This clause does not prevent the Developer and the Landowners from entering into variations of lease (including by way of exercise of option) or transfers of lease, over any part of the Land, provided that any variation of lease for any part of the Land on which this document is not registered does not extend the term of that lease (including any options) beyond its existing term, other than where either:
 - (i) the variation includes the clause contained in Permitted Leases; or
 - (ii) where the relevant land parcel is the subject of other leases, the variation would not extend the term of the lease in question beyond the term of the lease (including any options) with the longest lease period for that parcel of land.
- (g) Where there is a failure to Register this document on the title to any part of the Land within 90 Business Days of receiving a copy of this document executed by the Minister:
 - (i) the Developer and Landowners must comply with the obligation under **clause 8.1(d)** to continue to use all reasonable endeavours to procure Registration of the document on that part(s) of the Land as soon as possible;
 - (ii) the Minister may require provision of a Bank Guarantee or Bank Guarantees of up to \$1 million per Lot comprising the Land; and
 - (iii) the Developer must provide that Bank Guarantee or the Bank Guarantees to the Minister within 5 Business Days of any such request.

The Minister will return any Bank Guarantee or Bank Guarantees provided under this clause on confirmation of the Registration of this document on the title to the relevant part of the Land and **clause 1** of **Schedule 4** will apply to any Bank Guarantee provided under this clause.

8.2 Evidence of Registration

In respect of each part of the Land, the Developer and the Landowners must provide the Minister with a copy of the relevant folios of the Register and a copy of the Registered dealing within 10 Business Days of Registration of this document.

8.3 Right to lodge caveat

(a) Subject to **clause 8.3(b)**, until such time as this document is registered on the title of the Land, the Developer and the Landowners acknowledge that:

- (i) this document confers on the Minister, on and from the date of execution by the Parties, an interest in the Land and entitles the Minister to lodge and maintain a caveat on the title to the Land to prevent any Dealing in respect of the Land; and
- (ii) upon execution of this document, the Minister will lodge a caveat on the title to the Land, substantially in the form of the caveat attached at **Annexure E**.
- (b) If the Minister lodges a caveat in accordance with **clause 8.3(a)**, then the Minister:
 - (i) must do all things reasonably necessary to:
 - (A) ensure that the caveat does not prevent or delay the Registration of this document over any part of the Land;
 - (B) withdraw the caveat from the title to the relevant part of the Land promptly and within 5 Business Days of being provided with evidence of Registration of this document over that part of the Land in accordance with clause 8.1 and clause 8.2; and
 - (ii) must, without delay after being requested to do so, provide such consent or other documentation (such as a withdrawal of caveat) required to permit the Registration of:
 - (A) any easements to burden or benefit the Land;
 - (B) any variations of lease (including by way of exercise of option) subject to compliance with clause 8.1(f), or transfers of lease, over any part of the Land;
 - (C) any Permitted Lease;
 - (D) the subdivision of the Land for the purpose of creating parcels comprising the whole of each Contribution Area as identified on the Plan and any roads which are required to be created for the purpose of creating each Contribution Area; and
 - (E) the subdivision of Contribution Area 3 for the purpose of creating a separate title for the Education Land as identified on the Plan,
 - (iii) may, after being requested to do so, provide such consent or other documentation (such as a withdrawal of caveat) required to permit the Registration of any instrument recording Dealings in respect of any part of the Land. The Minister may reasonably refuse to provide such consent or other documentation to permit the Registration of any instrument recording Dealing until such time as either:

- (A) this document is Registered on the title to the relevant part of the Land, in which case the Minister will be required to withdraw the caveat under clause 8.3(b)(i)(B); or
- (B) all the Development Contributions have been provided in accordance with this document in respect of the relevant part of the Land.
- (c) The Developer and the Landowners must pay the Minister's reasonable costs and expenses, including legal costs, of exercising the Minister's rights under **clause 8.3(a)** to lodge a caveat(s) and withdraw caveat(s) and any associated costs.

8.4 Release and discharge of document

The Minister agrees to do all things reasonably required by the Developer and the Landowners to, following a request by the Developer or a relevant Landowner:

- (a) release and discharge this document with respect to the Land or any Contribution Area forming the subject of the request; and
- (b) execute the relevant documents to enable the relevant Landowner or Landowners to remove the notation of this document from the relevant folios of the Register in respect of the Land or relevant Contribution Area forming the subject of the request,

upon the Minister being satisfied that:

- (c) the Developer and the relevant Landowner or Landowners have satisfied all of their obligations under this document in respect of the Land or relevant Contribution Area forming the subject of the request;
- (d) the Developer and the relevant Landowner or Landowners are not otherwise in default of their obligations under this document in respect of any part of the relevant Contribution Area forming the subject of the request; and
- (e) if a request is made in respect of a Contribution Area that is owned by a Party that is a Landowner as at the date of execution of this document, the Developer and the relevant Landowner or Landowners are not in default of their obligations under this document in respect of the Education Land.

8.5 Landowner's interest in Land

Each Landowner represents and warrants separately that the relevant Landowner is:

- (a) the owner of the Land shown in **Schedule 2** other than the EnergyAustralia Land; and
- (b) legally entitled to obtain all consents and approvals and to compel any person referred to in or contemplated by **clause 8.1(b)(iii)** to assist, cooperate and otherwise do all things necessary for each Landowner to comply with its obligations under **clause 8**.

9 Dispute Resolution

9.1 Reference to dispute

If a dispute arises between the Parties in relation to this document, then either Party must resolve that dispute in accordance with this clause.

9.2 Notice of dispute

The Party wishing to commence the dispute resolution processes must notify the other in writing of:

- (a) the intent to invoke this clause;
- (b) the nature or subject matter of the dispute, including a summary of any efforts made to resolve the dispute other than by way of this clause; and
- (c) the outcomes which the notifying Party wishes to achieve (if practicable).

9.3 Representatives of Parties to meet

On receipt of notice under **clause 9.2**, the Parties must endeavour in good faith to resolve the dispute expeditiously using informal dispute resolution techniques such as mediation, expert evaluation or other techniques agreed by them.

9.4 Mediation

If the Parties do not agree within 20 Business Days of receipt of the notice under **clause 9.2** (or any further period agreed in writing by them) as to:

- (a) the dispute resolution techniques and procedures to be adopted;
- (b) the timetable for all steps in those procedures; or
- (c) the selection and compensation of the independent person required for such technique,

the Parties must mediate the dispute in accordance with the Mediation Rules of the Law Society of NSW. The Parties must request the president of the Law Society of NSW or the president's nominee to select the mediator and determine the mediator's remuneration.

9.5 Court proceedings

If the dispute is not resolved within 60 Business Days after notice is given under **clause 9.2** then any Party which has complied with the provisions of this **clause 9** may in writing terminate any dispute resolution process undertaken under this clause and may then commence court proceedings in relation to the dispute.

9.6 Not use information

The Parties acknowledge the purpose of any exchange of information or documents or the making of any offer of settlement under this **clause 9** is to attempt to settle the dispute. No Party may use information or documents obtained through any dispute resolution process undertaken under this **clause 9** for any purpose other than in an attempt to settle the dispute.

9.7 No prejudice

This **clause 9** does not prejudice the right of a Party to institute court proceedings for urgent injunctive or declaratory relief in relation to any matter arising out of or relating to this document.

10 Enforcement

10.1 Developer and Landowners to provide security

The following constitutes the security for the purposes of this document:

- (a) the registration of this document on the title to the Lots comprising the Land and the other security arrangements in **clause 8**;
- (b) clauses 3.2(c), 3.3 and 3.4 of Schedule 3; and
- (c) the security set out in Schedule 4.

10.2 Enforcement by any Party

- (a) Without limiting any other remedies available to the Parties, this document may be enforced by any Party in any court of competent jurisdiction. Nothing in this document prevents:
 - (i) a Party from bringing proceedings in the Land and Environment Court to enforce any aspect of this document or any matter to which this document relates: and
 - (ii) the Minister from exercising any function under the Act or any other Act or law relating to the enforcement of any aspect of this document or any matter to which this document relates.
- (b) The Developer and the Landowners indemnify the Minister from any Loss suffered or incurred in exercising the Minister's enforcement rights under clause 10.2(a), other than Loss that is suffered or incurred:
 - (i) as a result of the Minister bringing proceedings against another Party or Parties in a court of competent jurisdiction to enforce any aspect of this document or any matter to which this document relates; and
 - (ii) in circumstances where, and to the extent that, the court makes orders for the Minister to pay the other Party's or Parties' costs of those proceedings.

11 Notices

11.1 Delivery

Any notice, consent, information, application or request that must or may be given or made to a Party under this document is only given or made if it is in writing and:

(a) delivered or posted to that Party at its address set out below;

(b) faxed to that Party at its fax number set out below; or

(c) emailed to that Party at its email address set out below.

The Minister for Planning

Attention: Pip Stenekes

Address: Department of Planning and Environment

23-33 Bridge Street Sydney NSW 2000

Fax Number: 02 9228 6455

Email: pip.stenekes@planning.nsw.gov.au

Goodman Property Services (Aust) Pty Limited

Attention: Jason Little, General Manager

Address: Level 17, 60 Castlereagh Street

Sydney NSW 2000

Fax Number: 02 9230 7444

Email: Jason.Little@goodman.com

The Trust Company Limited

Attention: Jason Little, General Manager

Address: C/- Goodman Property Services (Aust) Pty

Limited

Level 17, 60 Castlereagh Street

Sydney NSW 2000

Fax Number: 02 9230 7444

Email: Jason.Little@goodman.com

Tallina Pty Ltd

Attention: Jason Little, General Manager

Address: C/- Goodman Property Services (Aust) Pty

Limited

Level 17, 60 Castlereagh Street

Sydney NSW 2000

Fax Number: 02 9230 7444

Email: Jason.Little@goodman.com

11.2 Change of details

If a Party gives the other Party three Business Days' notice of a change of its address, email address or fax number, any notice, consent, information, application or request is only given or made by that other Party if it is delivered, posted, emailed or faxed to the latest address, email address or fax number.

11.3 Giving of notice

Any notice, consent, information, application or request is to be treated as given or made at the following time:

- (a) If it is delivered, when it is left at the relevant address.
- (b) If it is sent by post, two Business Days after it is posted.
- (c) If it is sent by fax, as soon as the sender receives from the senders fax machine a report of an error free transmission to the correct fax number.
- (d) If it is emailed, at the time the email was sent, provided the email was received by the recipient.

11.4 Delivery outside of business hours

If any notice, consent, information, application or request is delivered, or an error free transmission report in relation to it is received, on a day that is not a business day, or if on a Business Day, after 5.00 pm on that day in the place of the Party to whom it is sent, it is to be treated as having been given or made at the beginning of the next Business Day.

12 Approvals and consent

Except as otherwise set out in this document, and subject to any statutory obligations, the Minister may give or withhold an approval or consent to be given under this document in the Minister's absolute discretion and subject to any conditions determined by the Minister. The Minister is not obliged to give reasons for giving or withholding consent or for giving consent subject to conditions.

13 Assignment and dealings

13.1 Restriction on assignment and novation

- (a) Neither the Developer nor the Landowners may:
 - (i) assign the rights or benefits; or
 - (ii) novate the burdens or obligations,
 - of this document to any person, unless the requirements of **clause 13.1(b)** are met.
- (b) The parties to any assignment or novation of any rights, benefits, burdens or obligations under this document must enter into a document

with the Minister substantially in the form of the Novation Deed. As a condition of entering into such a Novation Deed the Minister may require:

- (i) the provision of additional security required in addition to that identified in **clause 10**, provided that such additional security amount is no more than 10% of the Contribution Amount payable for the relevant Contribution Area and which amount is capped at \$1 million per Contribution Area; and
- (ii) an indemnity in favour of the Minister in respect of any breach of this document by the Developer or relevant Landowner.
- (c) The Minister may assign the rights and benefits of this document in the Minister's absolute discretion, without the need for consent from the Developer or the relevant Landowner.
- (d) For the avoidance of doubt and subject to this clause 13.1, a Landowner and the Developer may assign their respective rights and benefits and novate their respective burdens and obligations under this document as they relate to a Contribution Area. In such a case and subject to the terms of the Novation Deed, those rights, benefits, burdens and obligations will be severed as they relate to that Contribution Area. The rights, benefits, burdens and obligations of that Landowner and the Developer relating to the balance of the Land will not be affected.

13.2 Restriction on Dealings

- (a) This **clause 13.2** applies to any part of the Land where there has not been a release and discharge of this document under **clause 8.4**.
- (b) Neither the Developer nor the Landowners may complete any Dealing with any third party (**Transferee**) relating to their interests in the Land or any Contribution Area unless the Developer and Landowners have complied with:
 - (i) in relation to Contributions Areas 1, 2, 4 and 5, clauses 13.2(c) to
 (h) (as may be modified by clause 13.2(i) insofar as they relate to any Dealing in respect of Contribution Area 1); and
 - (ii) in relation to Contribution Area 3, clause 13.2(j).
- (c) Within 30 days of entering into any such Dealing, the Developer or relevant Landowner must give written notice to the Minister of:
 - (i) the existence of the Dealing;
 - (ii) the identity of the Transferee, and
 - (iii) the anticipated date of completion of the Dealing.
- (d) At least 50 days prior to the date proposed for completion of the Dealing, the Developer and the relevant Landowner must give to the Minister:
 - (i) written notice of their intention to complete the Dealing and of the proposed completion date (**Completion Date**);

- (ii) written notice of the identity of the Transferee (and whether that has changed from the Transferee identified in the notice given under clause 13.2(c)); and
- (iii) the completed novation deed substantially in the form of the Novation Deed which the parties to the Dealing intend to execute under which the Transferee agrees to comply with those obligations of the Developer or relevant Landowner (as applicable) under this document which relate to the relevant part of the Land the subject of the Dealing (**Required Obligations**) as if the Transferee were the Developer or relevant Landowner (as applicable) (**Prepared Novation Deed**).
- (e) At least 30 days prior to the Completion Date, the Minister must give to the Developer or relevant Landowner (as applicable) written notice of whether the Minister requires:
 - (i) any amendments to be made to the Prepared Novation Deed, provided that any such amendments must remain substantially in the form of the Novation Deed; and
 - (ii) as a condition of entering into such a Novation Deed, the provision of additional security required in addition to that identified in clauses 10 and 13, provided that such additional security amount is no more than 10% of the Contribution Amount payable for the relevant Contribution Area and which amount is capped at \$1 million per Contribution Area.
- (f) At least 20 days prior to the Completion Date, the Developer or relevant Land owner (as applicable) must give to the Minister the final form of the Prepared Novation Deed, incorporating the Minister's requested amendments under **clause 13.2(e)**, for the Minister's final review and approval, to which the Minister must respond within 7 days.
- (g) At least 5 Business Days prior to the Completion Date, the Developer or Landowner (as applicable) must deliver to the Minister the Novation Deed (incorporating any amendments required by the Minister under clause 13.2(e)) duly executed by the Developer, relevant Landowner, relevant Trustee and the Transferee.
- (h) Prior to the Completion Date:
 - (i) any default by the Developer or the relevant Landowner (as applicable) in relation to the relevant Contribution Area the subject of the proposed Dealing must be remedied by the Developer or the relevant Landowner (as applicable) unless that default has been waived expressly by the Minister; and
 - (ii) the Developer or the relevant Landowner (as applicable) and the Transferee must pay all costs reasonably incurred by the Minister in relation to that Dealing, including, without limitation, legal and administrative costs.

- (i) In the case of any Dealing in respect of Contribution Area 1, the application of this clause 13.2 is modified such that, in clause 13.2(d), the time specified in that clause shall be 14 days prior to the date for completion of the Dealing.
- (j) In relation to Contribution Area 3:
 - (i) until such time as Contribution Area 3 is subdivided to create a separate title for the Education Land, prior to entering into any Dealing with any Transferee relating to their interests in the Land relating to Contribution Area 3, the Developer, the relevant Landowner and the Transferee must enter into a Novation Deed in the form of **Annexure A** on terms acceptable to the Minister; and
 - (ii) at any time after a separate title has been created for the Education Land, the Developer, the relevant Landowner and the Transferee must comply with:
 - (A) **clause 13.2(j)(i)** in relation to any Dealing relating to their interests in the Education Land; and
 - (B) **clause 13.2(c)** to **clause 13.2(h)** in relation to any Dealing with any Transferee relating to their interests in the Land comprising the remainder of Contribution Area 3.
- (k) No later than one Business Day prior to the Completion Date, the Developer and relevant Landowner must provide to the Minister a Bank Guarantee for any additional security required by the Minister as specified in **clause 13.2(e)**.
- (I) As soon as reasonably practicable after receiving the executed Novation Deed and any Bank Guarantee referred to in **clause 13.2(e)** or any executed Novation Deed pursuant to **clause 13.2(j)**, the Minister must:
 - (i) execute the Novation Deed and deliver counterparts to the Developer or Landowner (as applicable);
 - (ii) return to the Developer and relevant Landowner their Bank Guarantee or Bank Guarantees referred to in clause 1.2(a) of Schedule 4 in relation to each Contribution Area which is the subject of the Dealing; and
 - (iii) if requested by the Developer and relevant Landowner:
 - (A) accept in exchange for the Bank Guarantee referred to in clause 13.2(e) a substitute Bank Guarantee provided by the Transferee for the same amount; and
 - (B) return the Bank Guarantee referred to in **clause 13.2(e)** to the Developer and relevant Landowner.
- (m) If the Developer or a Landowner enters into any Dealing and fully satisfies the requirements of **clause 13.2(b)** to **clause 13.2(j)**, as relevant with respect to that Dealing, the Developer or the relevant Landowner (as applicable) will be released from its obligations under this

- document with respect to the Required Obligations as they relate to the Land the subject of that Dealing.
- (n) For the avoidance of doubt and subject to this clause 13.2, the Developer or a Landowner may novate its obligations under this document as they relate to a Contribution Area. In such a case and subject to the terms of the Novation Deed, those obligations will be severed as they relate to that Contribution Area. The obligations of the Developer or a Landowner relating to the balance of the Land will not be affected.

14 Capacity

- (a) Each Party warrants to each other Party that this document creates legal, valid and binding obligations, enforceable against the relevant Party in accordance with its terms.
- (b) If an attorney executes this document on behalf of any Party, the attorney declares that it has no notice of the revocation of that power of attorney.

15 Reporting requirement

- (a) The relevant Landowner must provide at least 20 Business Days notice to the Minister of any application made by the Landowner or any other third party (including the Developer) in relation to the Land for Planning Approval, a Complying Development Certificate, a Construction Certificate or a Subdivision Certificate which triggers any obligation under this document.
- (b) If any part of the Development is to be carried out without the need for Planning Approval and which triggers any obligation under this document, then the relevant Landowner must provide at least 20 Business Days notice to the Minister of the date of the proposed commencement of the relevant part of the Development.
- (c) On each anniversary of the date of this document or as otherwise agreed with the Secretary, the Landowners must deliver to the Secretary a report which must include those matters set out in clause 15(d) and clause 15(e), as applicable.
- (d) If neither the Developer nor the relevant Landowner has provided a Contribution Amount in the 12 month period immediately preceding the relevant anniversary of this document, the report must include:
 - (i) a description of the status of the Development;
 - (ii) a forecast in relation to the anticipated progression and completion of the Development; and

- (iii) an estimated date for when the Developer, the relevant Landowner or any other third party (as applicable) expects to lodge the first Planning Application and any application for a Complying Development Certificate, Construction Certificate or Subdivision Certificate for the relevant part of the Land.
- (e) If the Developer and the Landowners have provided one or more Contribution Amounts under this document, the report must include:
 - details of all Planning Approvals, Complying Development Certificates, Construction Certificates and Subdivision Certificates granted in relation to the Development;
 - (ii) a schedule that details all Contribution Amounts provided under this document as at the date of the report; and
 - (iii) an estimated date for when the Developer, the relevant Landowner or any other third party (as applicable) expects to lodge the next Planning Application and application for any Complying Development Certificate, Construction Certificate or Subdivision Certificate for the relevant part of the Land.
- (f) Upon the Secretary's request, the Developer and the relevant Landowner must deliver to the Secretary all documents and other information which, in the reasonable opinion of the Secretary, are necessary for the Secretary to assess the status of the Development.

16 Limitation of liability for the Carter Street Trust

The provisions of this **clause 16** apply despite anything to the contrary in this document.

16.1 Definitions

In this clause 16:

- (a) **Assets** includes all assets, property and rights, real and personal, of any value whatsoever, of the Trust.
- (b) **Constitution** means the constitution of the Trust as amended from time to time.
- (c) **Custody Agreement** means the Carter Street Custody Agreement dated 11 February 2002.
- (d) Obligations means all obligations and liabilities of whatever kind undertaken or incurred by, or devolving upon, the First Landowner under or in respect of this document.
- (e) **Tallina** means Tallina Pty Ltd ACN 090 716 895.
- (f) Trust means the Carter Street Trust ABN 12 909 843 665.
- (g) **Trustee** means the entity from time to time acting in the Trustee's Capacity, which on the making of this document is Tallina.

(h) **Trustee's Capacity** means the capacity in which the Trustee enters into this document, being as trustee of the Trust.

16.2 Limitation of the First Landowner's Liability

- (a) The First Landowner enters into this document as custodian for the Trust and in no other capacity.
- (b) The parties other than the First Landowner acknowledge that the Obligations are incurred by the First Landowner solely in its capacity as custodian of the Assets and that the First Landowner will cease to have any Obligation under this document if the First Landowner ceases for any reason to be owner of the Assets.
- (c) The First Landowner will not be liable to pay or satisfy any Obligations except to the extent to which it is indemnified or entitled to be indemnified:
 - (i) by the Trustee; or
 - (ii) out of the Assets in respect of any liability incurred by it.
- (d) The parties other than the First Landowner may enforce their rights against the First Landowner arising from non-performance of the Obligations only to the extent of the First Landowner indemnities referred to in clause 16.2(c).
- (e) Subject to **clause 16.2(e)(i)**, if any party other than the First Landowner does not recover all money owing to it arising from non-performance of the Obligations it may not seek to recover the shortfall by:
 - (i) bringing proceedings against the First Landowner in its personal capacity; or
 - (ii) applying to have the First Landowner wound up or proving in the winding up of the First Landowner.
- (f) The provisions of this **clause 16.2** do not apply to any obligation or liability of the First Landowner to the extent that it is not satisfied because, under the Custody Agreement or by operation of law, there is a reduction in the extent of the First Landowner's indemnification from the Trustee or out of the Assets as a result of the First Landowner's fraud, negligence or breach of duty.
- (g) Except in the case of and to the extent of fraud, negligence or breach of duty on the part of the First Landowner under its custody agreement with the Trustee, the parties other than the First Landowner waive their rights and release the First Landowner from any personal liability whatsoever, in respect of any Loss or damage which they may suffer as a result of any:
 - (i) breach by the First Landowner of any of its Obligations; or
 - (ii) non-performance by the First Landowner of the Obligations but only to the extent to which the liability cannot be paid or satisfied

by the indemnities set out above in **clause 16.2(c)** in respect of any liability incurred by it.

The parties other than The First Landowner acknowledge that the whole of this document is subject to this clause.

- (h) The parties acknowledge that the Trustee is responsible under the Constitution for performing a variety of obligations relating to the Trust, including under this document. The parties agree that no act or omission of the First Landowner (including any related failure to satisfy any Obligations) will constitute fraud, negligence or breach of duty of the First Landowner for the purposes of clause 16.2(e)(i) to the extent to which the act or omission was caused or contributed to by any failure of the Trustee or any other person to fulfil its obligations relating to the Trust or by any other act or omission of the Trustee or any other person.
- (i) No attorney, agent or other person appointed in accordance with this document has authority to act on behalf of the First Landowner in a way which exposes the First Landowner to any personal liability and no act or omission of such a person will be considered fraud, negligence or breach of duty of the First Landowner for the purposes of clause 16.2(e)(i).

16.3 Limitation of Trustee's Liability

(a) Capacity

The Trustee's liability under this document is limited to the Trustee's Capacity and the Trustee is not liable in any other capacity.

(b) Limitation

Subject to **clause 16.3(d)**, the liability of the Trustee in respect of any cause of action, Claim or Loss arising:

- (i) under or in connection with this document;
- (ii) in connection with any transaction, conduct or any other agreement contemplated by this document; or
- (iii) under or in connection with (to the extent permitted by law) any representation or undertaking given or to be given in connection with this document,

(each, a **Trust Claim**) and the right of each party other than the Trustee to recover from the Trustee any amount in respect of any (and all) Trust Claims is limited to a right to recover an amount not exceeding the amount which the Trustee is entitled and able to recover from the Assets (after taking account of the costs of exercising its right of indemnity or exoneration) and if, after exercise of those rights, any such amount remains outstanding, no further Trust Claim may be made against the Trustee personally in respect of that outstanding amount.

(c) Acknowledgment of limitations

The parties other than the Trustee agree and acknowledge that they must not, in respect of any Trust Claim:

- (i) subject to **clause 16.3(d)**, bring proceedings against the Trustee in its personal capacity;
- (ii) seek to appoint an administrator or liquidator to the Trustee;
- (iii) commence the winding-up, dissolution or administration of the Trustee; or
- (iv) appoint a receiver, receiver and manager, administrative receiver or similar official to all or any of the assets of the Trustee,

except to the extent that the steps taken affect any Assets or the Trustee's right of recourse against, and indemnity from, the Assets and nothing else.

(d) Exception

If the Trustee acts negligently, fraudulently, with wilful misconduct or in breach of trust with a result that:

- (i) the Trustee's right of indemnity, exoneration or recoupment of the Assets; or
- (ii) the actual amount recoverable by the Trustee in exercise of those rights,

is reduced in whole or in part or does not exist, then to the extent that such right or the amount so recoverable is reduced or does not exist, the Trustee will be personally liable.

17 Limitation of liability for the Hill Road Trust

The provisions of this **clause 17** apply despite anything to the contrary in this document.

17.1 Definitions

In this clause 17:

- (a) **Assets** includes all assets, property and rights, real and personal, of any value whatsoever, of the Trust.
- (b) **Constitution** means the constitution of the Trust as amended from time to time.
- (c) **Custody Agreement** means the Hill Road Trust Custody Agreement dated 30 October 2002.
- (d) **Obligations** means all obligations and liabilities of whatever kind undertaken or incurred by, or devolving upon, the Second Landowner under or in respect of this document.

Tallina means Tallina Pty Ltd ACN 090 716 895.

(e) Trust means the Hill Road Trust ABN 54 244 152 320.

- (f) **Trustee** means the entity from time to time acting in the Trustee's Capacity which on the making of this document is Tallina.
- (g) **Trustee's Capacity** means the capacity in which the Trustee enters into this document, being as trustee of the Trust.

17.2 Limitation of the Second Landowner's Liability

- (a) The Second Landowner enters into this document as custodian for the Trust and in no other capacity.
- (b) The parties other than the Second Landowner acknowledge that the Obligations are incurred by the Second Landowner solely in its capacity as custodian of the Assets and that the Second Landowner will cease to have any Obligation under this document if the Second Landowner ceases for any reason to be owner of the Assets.
- (c) The Second Landowner will not be liable to pay or satisfy any Obligations except to the extent to which it is indemnified or entitled to be indemnified:
 - (i) by the Trustee; or
 - (ii) out of the Assets in respect of any liability incurred by it.
- (d) The parties other than the Second Landowner may enforce their rights against the Second Landowner arising from non-performance of the Obligations only to the extent of the Second Landowner indemnities referred to in clause 17.2(c).
- (e) Subject to **clause 17.2(e)(i)**, if any party other than the Second Landowner does not recover all money owing to it arising from non-performance of the Obligations it may not seek to recover the shortfall by:
 - (i) bringing proceedings against the Second Landowner in its personal capacity; or
 - (ii) applying to have the Second Landowner wound up or proving in the winding up of the Second Landowner.
- (f) The provisions of this **clause 17.2** do not apply to any obligation or liability of the Second Landowner to the extent that it is not satisfied because, under the Custody Agreement or by operation of law, there is a reduction in the extent of the Second Landowner's indemnification from the Trustee or out of the Assets as a result of the Second Landowner's fraud, negligence or breach of duty.
- (g) Except in the case of and to the extent of fraud, negligence or breach of duty on the part of the Second Landowner under its custody agreement with the Trustee, the parties other than the Second Landowner waive their rights and release the Second Landowner from any personal liability whatsoever, in respect of any Loss or damage which they may suffer as a result of any:
 - (i) breach by the Second Landowner of any of its Obligations; or

(ii) non-performance by the Second Landowner of the Obligations, but only to the extent to which the liability cannot be paid or satisfied by the indemnities set out above in clause 17.2(c) in respect of any liability incurred by it.

The parties other than The Second Landowner acknowledge that the whole of this document is subject to this clause.

- (h) The parties acknowledge that the Trustee is responsible under the Constitution for performing a variety of obligations relating to the Trust, including under this document. The parties agree that no act or omission of the Second Landowner (including any related failure to satisfy any Obligations) will constitute fraud, negligence or breach of duty of the Second Landowner for the purposes of clause 17.2(e)(i) to the extent to which the act or omission was caused or contributed to by any failure of the Trustee or any other person to fulfil its obligations relating to the Trust or by any other act or omission of the Trustee or any other person.
- (i) No attorney, agent or other person appointed in accordance with this document has authority to act on behalf of the Second Landowner in a way which exposes the Second Landowner to any personal liability and no act or omission of such a person will be considered fraud, negligence or breach of duty of the Second Landowner for the purposes of clause 17.2(e)(i).

17.3 Limitation of Trustee's Liability

(a) Capacity

The Trustee's liability under this document is limited to the Trustee's Capacity and the Trustee is not liable in any other capacity.

(b) Limitation

Subject to **clause 17.3(d)**, the liability of the Trustee in respect of any cause of action, Claim or Loss arising:

- (i) under or in connection with this document;
- (ii) in connection with any transaction, conduct or any other agreement contemplated by this document; or
- (iii) under or in connection with (to the extent permitted by law) any representation or undertaking given or to be given in connection with this document.

(each, a **Trust Claim**) and the right of each party other than the Trustee to recover any amount from the Trustee in respect of any (and all) Trust Claims is limited to a right to recover an amount not exceeding the amount which the Trustee is entitled and able to recover from the Assets (after taking account of the costs of exercising its right of indemnity or exoneration) and if, after exercise of those rights, any such amount remains outstanding, no further Trust Claim may be made against the Trustee personally in respect of that outstanding amount.

(c) Acknowledgment of limitations

The parties other than the Trustee agree and acknowledge that they must not, in respect of any Trust Claim:

- (i) subject to **clause 17.3(d)**, bring proceedings against the Trustee in its personal capacity;
- (ii) seek to appoint an administrator or liquidator to the Trustee;
- (iii) commence the winding-up, dissolution or administration of the Trustee; or
- (iv) appoint a receiver, receiver and manager, administrative receiver or similar official to all or any of the assets of the Trustee,

except to the extent that the steps taken affect any Assets or the Trustee's right of recourse against, and indemnity from, the Assets and nothing else.

(d) Exception

If the Trustee acts negligently, fraudulently, with wilful misconduct or in breach of trust with a result that:

- (i) the Trustee's right of indemnity, exoneration or recoupment of the Assets; or
- (ii) the actual amount recoverable by the Trustee in exercise of those rights,

is reduced in whole or in part or does not exist, then to the extent that such right or the amount so recoverable is reduced or does not exist, the Trustee will be personally liable.

18 Limitation of liability for the IBC Trust

The provisions of this **clause 18** apply despite anything to the contrary in this document.

18.1 Definitions

In this clause 18:

- (a) **Assets** includes all assets, property and rights, real and personal, of any value whatsoever, of the Trust.
- (b) **Constitution** means the constitution of the Trust as amended from time to time.
- (c) Custody Agreement means the IBC Trust Custody Agreement dated 11 May 2005.
- (d) Obligations means all obligations and liabilities of whatever kind undertaken or incurred by, or devolving upon, the Third Landowner under or in respect of this document.

- (e) **Tallina** means Tallina Pty Ltd ACN 090 716 895.
- (f) **Trust** means the IBC Trust ABN 30 476 081 229.
- (g) **Trustee** means the entity from time to time acting in the Trustee's Capacity which on the making of this document is Tallina.
- (h) **Trustee's Capacity** means the capacity in which the Trustee enters into this document, being as trustee of the Trust.

18.2 Limitation of the Third Landowner's Liability

- (a) The Third Landowner enters into this document as custodian for the Trust and in no other capacity.
- (b) The parties other than the Third Landowner acknowledge that the Obligations are incurred by the Third Landowner solely in its capacity as custodian of the Assets and that the Third Landowner will cease to have any Obligation under this document if the Third Landowner ceases for any reason to be owner of the Assets.
- (c) The Third Landowner will not be liable to pay or satisfy any Obligations except to the extent to which it is indemnified or entitled to be indemnified:
 - (i) by the Trustee; or
 - (ii) out of the Assets in respect of any liability incurred by it.
- (d) The parties other than the Third Landowner may enforce their rights against the Third Landowner arising from non-performance of the Obligations only to the extent of the Third Landowner indemnities referred to in clause 18.2(c).
- (e) Subject to **clause 18.2(e)(i)**, if any party other than the Third Landowner does not recover all money owing to it arising from non-performance of the Obligations it may not seek to recover the shortfall by:
 - (i) bringing proceedings against the Third Landowner in its personal capacity; or
 - (ii) applying to have the Third Landowner wound up or proving in the winding up of the Third Landowner.
- (f) The provisions of this **clause 18.2** do not apply to any obligation or liability of the Third Landowner to the extent that it is not satisfied because, under the Custody Agreement or by operation of law, there is a reduction in the extent of the Third Landowner's indemnification from the Trustee or out of the Assets as a result of the Third Landowner's fraud, negligence or breach of duty.
- (g) Except in the case of and to the extent of fraud, negligence or breach of duty on the part of the Third Landowner under its custody agreement with the Trustee, the parties other than the Third Landowner waive their rights and release the Third Landowner from any personal liability

whatsoever, in respect of any Loss or damage which they may suffer as a result of any:

- (i) breach by the Third Landowner of any of its Obligations; or
- (ii) non-performance by the Third Landowner of the Obligations but only to the extent to which the liability cannot be paid or satisfied by the indemnities set out above in clause 18.2(c) in respect of any liability incurred by it.

The parties other than The Third Landowner acknowledge that the whole of this document is subject to this clause.

- (h) The parties acknowledge that the Trustee is responsible under the Constitution for performing a variety of obligations relating to the Trust, including under this document. The parties agree that no act or omission of the Third Landowner (including any related failure to satisfy any Obligations) will constitute fraud, negligence or breach of duty of the Third Landowner for the purposes of clause 18.2(e)(i) to the extent to which the act or omission was caused or contributed to by any failure of the Trustee or any other person to fulfil its obligations relating to the Trust or by any other act or omission of the Trustee or any other person.
- (i) No attorney, agent or other person appointed in accordance with this document has authority to act on behalf of the Third Landowner in a way which exposes the Third Landowner to any personal liability and no act or omission of such a person will be considered fraud, negligence or breach of duty of the Third Landowner for the purposes of clause 18.2(e)(i).

18.3 Limitation of Trustee's Liability

(a) Capacity

The Trustee's liability under this document is limited to the Trustee's Capacity and the Trustee is not liable in any other capacity.

(b) Limitation

Subject to **clause 18.3(d)**, the liability of the Trustee in respect of any cause of action, Claim or Loss arising:

- (i) under or in connection with this document;
- (ii) in connection with any transaction, conduct or any other agreement contemplated by this document; or
- (iii) under or in connection with (to the extent permitted by law) any representation or undertaking given or to be given in connection with this document,

(each, a **Trust Claim**) and the right of each party other than the Trustee to recover any amount from the Trustee in respect of any (and all) Trust Claims is limited to a right to recover an amount not exceeding the amount which the Trustee is entitled and able to recover from the Assets (after taking account of the costs of exercising its right of indemnity or

exoneration) and if, after exercise of those rights, any such amount remains outstanding, no further Trust Claim may be made against the Trustee personally in respect of the outstanding amount.

(c) Acknowledgment of limitations

The parties other than the Trustee agree and acknowledge that they must not, in respect of any Trust Claim:

- (i) subject to **clause 18.3(d)**, bring proceedings against the Trustee in its personal capacity;
- (ii) seek to appoint an administrator or liquidator to the Trustee;
- (iii) commence the winding-up, dissolution or administration of the Trustee; or
- (iv) appoint a receiver, receiver and manager, administrative receiver or similar official to all or any of the assets of the Trustee,

except to the extent that the steps taken affect any Assets or the Trustee's right of recourse against, and indemnity from, the Assets and nothing else.

(d) Exception

If the Trustee acts negligently, fraudulently, with wilful misconduct or in breach of trust with a result that:

- (i) the Trustee's right of indemnity, exoneration or recoupment of the Assets; or
- (ii) the actual amount recoverable by the Trustee in exercise of those rights,

is reduced in whole or in part or does not exist, then to the extent that such right or the amount so recoverable is reduced or does not exist, the Trustee will be personally liable.

19 Trustee's representations and warranties

19.1 Definitions

In this clause 19:

- (a) **Assets** includes all assets, property and rights, real and personal, of any value whatsoever, of the Trusts.
- (b) **Beneficiaries** means the beneficiaries of the Trusts.
- (c) **Custody Agreements** means the following:
 - (i) the IBC Trust Custody Agreement dated11 May 2005;
 - (ii) the Hill Road Trust Custody Agreement dated 30 October 2002; and
 - (iii) the Carter Street Custody Agreement dated 11 February 2002.

- (d) Obligations means all obligations and liabilities of whatever kind undertaken or incurred by, or devolving upon, the First Landowner, the Second Landowner and the Third Landowner under or in respect of this document.
- (e) **Trusts** means the Carter Street Trust ABN 12 909 843 665, the Hill Road Trust ABN 54 244 152 320 and the IBC Trust 30 476 081 229.
- (f) Trust Deeds means:
 - (i) the IBC Trust Constitution dated 11 May 2005 between the Trustee and the relevant unit holders;
 - (ii) the Hill Road Trust Constitution dated 30 October 2002 between the Trustee and the relevant unit holders; and
 - (iii) the Carter Street Trust Constitution dated 11 February 2002 between the Trustee and the relevant unit holders.
- (g) **Trustee** means the entity from time to time acting in the Trustee's Capacity, which on the making of this document is Tallina.
- (h) **Trustee's Capacity** means the capacity in which the Trustee enters into this document, being as trustee of the Trust.
- (i) **Trustee Documents** means this document and any documents created under or in connection with this document.

19.2 The Trustee's representations and warranties

The Trustee represents and warrants to the Minister that:

- (a) powers of Trustee and custodian:
 - (i) the Trustee has power as trustee of the Trusts to execute the Trustee Documents; and
 - (ii) each Landowner has power as custodian of the relevant Trusts to execute the Trustee Documents:
- (b) **due performance:** in executing the Trustee Documents, the Trustee and its representatives have properly performed their obligations to the Beneficiaries:
- (c) **Trustee action taken:** all necessary action required by the Trust Deeds and the Custody Agreement to authorise the unconditional execution and delivery of, and compliance with the Trustee's obligations under, the Trustee Documents has been taken;
- (d) **custodian action taken:** all necessary action required by the Trust Deeds and the Custody Agreement to authorise the unconditional execution and delivery of, and compliance with each of the Landowner's obligations under, the Trustee Documents has been taken;
- (e) **Trustee instructions** the Trustee will give all necessary instructions to all Landowners to enable the Landowners to execute and deliver, and comply with all of their obligations under, the Trustee Documents;

(f) sole Trustee and custodian:

- (i) the Trustee is the only trustee of the Trusts; and
- (ii) the Landowners are the only custodians of the relevant Trusts;
- (g) **no removal of Trustee:** no action has been taken to remove the Trustee as trustee of the Trusts or to appoint an additional trustee of the Trusts;
- (h) **no removal of custodian:** no action has been taken to remove any or all of the Landowners as custodians of the relevant Trusts or to appoint any additional custodian of the relevant Trusts;

(i) Trustee's right of indemnity:

- (i) the Trustee has a right to be fully indemnified out of the Assets;
- (ii) the Trustee has not released or disposed of the Trustee's equitable lien over the Assets which secures that indemnity;
- (iii) the Trustee has not committed any breach of trust or done or omitted to do anything which has prejudiced or limited its rights of indemnity or equitable lien; and
- (iv) the Assets are sufficient to satisfy that indemnity;

(j) custodian's right of indemnity:

- (i) each Landowner has a right to be fully indemnified from the Assets and by the Trustee for all liabilities under the Trustee Documents;
- (ii) the Landowners have not released or disposed of the Landowners' equitable lien over the Assets which secures that indemnity;
- (iii) the Landowners have not committed any breach of the Custody Agreement or done or omitted to do anything which has prejudiced or limited their rights of indemnity or equitable lien; and
- (iv) the Assets are sufficient to satisfy that indemnity;

(k) no default:

- (i) the Trustee has not defaulted in the performance and observance of its obligations as trustee of the Trusts; and
- (ii) so far as the Trustee is aware, the Landowners have not defaulted in the performance and observance of their obligations as custodians of the Trusts or under the Custody Agreement so far as the Trustee is aware, the Landowners have not defaulted in the performance and observance of their obligations as custodians of the Trusts or under the Custody Agreement;
- (I) **Trusts or Custody Agreement not terminated:** no action has been taken or, so far as the Trustee is aware, is contemplated to terminate the Trusts or the Custody Agreement;
- (m) **full disclosure:** the Trustee has disclosed to the Minister full particulars of the Trusts, the Custody Agreement and any other trust or fiduciary relationship affecting the Assets and, without limitation, has given to the

- Minister copies of any instruments creating or evidencing the Trusts and the Custody Agreement;
- (n) **Trusts duly constituted:** the Trusts are duly constituted and are not void, voidable or otherwise unenforceable;
- (o) **custodian duly appointed:** each Landowner has been duly appointed as custodian of the relevant Trusts;
- (p) capacity of Trustee: the Trustee:
 - is duly incorporated in accordance with the laws of its place of incorporation as stated in this document, validly exists under those laws and has the capacity to sue and be sued in its own name, to own property and to act as trustee of the Trusts; and
 - (ii) is not insolvent.

19.3 Trustee's additional obligations

- (a) The Trustee must:
 - ensure the Assets are not mixed with any other property;
 - (ii) comply with the Trustee's obligations as trustee of the Trusts;
 - (iii) not release, dispose of or otherwise prejudice the Trustee's right of indemnity against, and equitable lien over, the Assets and the Trustee's right of indemnity (if any) against the Beneficiaries;
 - (iv) at the Minister's request, exercise the Trustee's right of indemnity against, and equitable lien over, the Assets and the Trustee's right of indemnity (if any) against the Beneficiaries; and
 - (v) not, without obtaining the Minister's prior written approval:
 - (A) terminate any or all of the Trusts;
 - (B) resettle the Assets; or
 - (C) vary the terms of the Trust if to do so would adversely affect the Trustee's right of indemnity against, or equitable lien over, the Assets or the Trustee's right of indemnity (if any) against the Beneficiaries.
- (b) The Trustee covenants that it will not retire or be replaced as trustee of any or all of the Trusts unless:
 - (i) another entity has been appointed in its place for the relevant Trusts:
 - (ii) the new entity agrees to be bound by the Trustee Documents, the Trust Deeds and the Custody Agreements (as applicable) as trustee for the relevant parcel of Land; and
 - (iii) unless the new entity is a Developer-Related Entity, the Trustee obtains the Minister's prior written consent, which must not be unreasonably withheld, to the appointment of the new entity.

19.4 The Landowners' representations and warranties

Each Landowner represents and warrants to the Minister that:

- (a) **custodian's power:** the Landowner has power as custodian of the relevant Trusts to execute the Trustee Documents;
- (b) **due performance:** in executing the Trustee Documents, the Landowner and its representatives have properly performed their obligations to the Beneficiaries and Trustee;
- (c) **custodian action taken:** all necessary action required by the Trust Deeds and the Custody Agreement to authorise the unconditional execution and delivery of, and compliance with the Landowner's obligations under, the Trustee Documents has been taken;
- (d) **no removal of custodian:** so far as the Landowner is aware, no action has been taken to remove the Landowner as custodian of the relevant Trusts or to appoint an additional custodian of the relevant Trusts;
- (e) **custodian duly appointed:** each Landowner has been duly appointed as custodian of the relevant Trusts;
- (f) capacity of custodian: the Landowner:
 - (i) is duly incorporated in accordance with the laws of its place of incorporation as stated in this document, validly exists under those laws and has the capacity to sue and be sued in its own name, to own property and to act as custodian of the Trusts; and
 - (ii) is not insolvent.

19.5 The Landowners' and Trustees' additional obligations

- (a) Each Landowner must:
 - (i) ensure the Assets are not mixed with any other property;
 - (ii) comply with the its obligations under the Custody Agreements as custodian of the relevant Trusts;
 - (iii) not release, dispose of or otherwise prejudice the Landowner's right of indemnity against, and equitable lien over, the Assets and the Landowner's right of indemnity (if any) against the Beneficiaries:
 - (iv) at the Minister's request, exercise the Landowner's right of indemnity against, and equitable lien over, the Assets and the Landowner's right of indemnity (if any) against the Beneficiaries;
- (b) Each Landowner covenants that it will not retire or be replaced (and each Trustee covenants that it will not remove the Landowner or cause the Landowner to retire) as custodian of any or all of the Trusts unless:
 - (i) another entity has been appointed in its place for the relevant Trusts; and

- (ii) the new entity agrees to be bound by the Trustee Documents, the Trust Deeds and the Custody Agreements (as applicable) as Landowner for the relevant parcel of Land.
- (c) No Landowner may take any action to amend the Trust Deeds or the Custody Agreements without the prior written consent of the Minister, acting reasonably.

19.6 Non Merger

This **clause 19** does not merge on the completion, rescission or other termination of this document or on the provision of any Development Contributions under this document.

20 GST

20.1 Construction

In this clause 20:

- (a) unless there is a contrary indication, words and expressions which are not defined in this document but which have a defined meaning in the GST Law have the same meaning as in the GST Law; and
- (b) references to GST payable and input tax credit entitlements include:
 - (i) notional GST payable by, and notional input tax credit entitlements of the Commonwealth, a State or a Territory (including a government, government body, authority, agency or instrumentality of the Commonwealth, a State or a Territory); and
 - (ii) GST payable by, and the input tax credit entitlements of, the representative member of a GST group of which the entity is a member.

20.2 Consideration GST exclusive

Unless otherwise expressly stated, all consideration, whether monetary or non-monetary, payable or to be provided to the Minister under or in connection with this document is exclusive of GST (**GST- exclusive consideration**).

20.3 Payment of GST

If GST is payable on any supply made by:

- (a) a party; or
- (b) an entity that is taken under the GST Law to make the supply by reason of the capacity in which a party acts,

(**Supplier**) under or in connection with this document, the recipient of the supply, or the party providing the consideration for the supply, must pay to the Supplier an amount equal to the GST payable on the supply.

20.4 Timing of GST payment

The amount referred to in **clause 20.3** must be paid in addition to and at the same time and in the same manner (without any set-off or deduction) that the GST-exclusive consideration for the supply is payable or to be provided.

20.5 Tax invoice

The Supplier must deliver a tax invoice or an adjustment note to the recipient of a taxable supply before the Supplier is entitled to payment of an amount under clause 20.3.

20.6 Adjustment event

If an adjustment event arises in respect of a supply made by a Supplier under or in connection with this document, any amount that is payable under **clause 20.3** will be calculated or recalculated to reflect the adjustment event and a payment will be made by the recipient to the Supplier or by the Supplier to the recipient as the case requires.

20.7 Reimbursements

- (a) Where a Party is required under or in connection with this document to pay for, reimburse or contribute to any expense, loss, liability or outgoing suffered or incurred by another party or indemnify another party in relation to such an expense, loss, liability or outgoing (**Reimbursable Expense**), the amount required to be paid, reimbursed or contributed by the first party will be reduced by the amount of any input tax credits to which the other party is entitled in respect of the Reimbursable Expense.
- (b) This **clause 20.7** does not limit the application of **clause 20.3**, if appropriate, to the Reimbursable Expense as reduced in accordance with **clause 20.7(a)**.

20.8 Calculations based on other amounts

If an amount of consideration payable or to be provided under or in connection with this document is to be calculated by reference to:

- (a) any expense, loss, liability or outgoing suffered or incurred by another person (Cost), that reference will be to the amount of that Cost excluding the amount of any input tax credit entitlement of that person relating to the Cost suffered or incurred; and
- (b) any price, value, sales, proceeds, revenue or similar amount (**Revenue**), that reference will be to that Revenue determined by deducting from it an amount equal to the GST payable on the supply for which it is consideration.

20.9 No merger

This **clause 20** does not merge on the completion, rescission or other termination of this document or on the transfer of any property supplied under this document.

21 General provisions

21.1 Costs, expenses and stamp duty

- (a) The Developer and the Landowners must pay their own and the Minister's reasonable legal costs and disbursements in connection with the negotiation, preparation, execution and carrying into effect of this document.
- (b) The Developer and the Landowners must pay all reasonable costs and expenses associated with the giving of public notice of this document and the Explanatory Note in accordance with the Regulation.
- (c) The Developer and the Landowners must pay all Taxes assessed on or in respect of this document and any instrument or transaction required or contemplated by or necessary to give effect to this document (including, but not limited to, stamp duty and Registration fees, if applicable, on any transfer of the Education Land).
- (d) The Developer and the Landowners must provide the Minister with bank cheques in respect of any costs incurred by the Minister pursuant to clauses 21.1(a), 21.1(b) and 21.1(c):
 - (i) where the Minister has provided the Developer or relevant Landowner written notice of the sum of such costs prior to execution, on the date of execution of this document; or
 - (ii) where the Minister has not provided the Developer or relevant Landowner with prior written notice of the sum of such costs prior to execution, within 30 Business Days of demand by the Minister for payment.

21.2 Entire agreement

- (a) This document contains everything to which the Parties have agreed in relation to the matters it deals with. No Party can rely on an earlier document, or anything said or done by another Party, or by a director, officer, agent or employee of that Party, before this document was executed, except as permitted by law.
- (b) Pursuant to clause 25E(7) of the Regulation, the Parties agree that the Explanatory Note is not to be used to assist in construing this document.

21.3 Variation

- (a) This document must not be varied except by a later written document executed by all Parties.
- (b) Subject to clause 21.3(c), the Parties agree that the requirements of clause 21.3(a) will be satisfied where the Minister, in the Minister's absolute discretion, agrees to the Developer or the relevant Landowner making minor amendments relating to the delivery of the Development and those amendments, and the consequences of those amendments under this document, have been agreed by the Parties through an

exchange of letters signed by or on behalf of the contact for each Party specified in **clause 11** of this document.

(c) Clause 21.3(b):

- (i) does not apply where the Act requires public notification of any amendment to this document;
- (ii) is subject to the Developer or the relevant Landowner obtaining all required approvals including any Planning Approval or modification to a Planning Approval; and
- (iii) is subject to **clause 21.7** of this document.

21.4 Further acts

Each Party must promptly execute all documents and do all things that another Party from time to time reasonably requests to affect, perfect or complete this document and all transactions incidental to it.

21.5 Governing law and jurisdiction

This document and the transactions contemplated by this document are governed by and are to be construed in accordance with the laws applicable in New South Wales. The Parties irrevocably and unconditionally submit to the non-exclusive jurisdiction of its courts and courts of appeal from them. The Parties will not object to the exercise of jurisdiction by those courts on any basis.

21.6 Joint and individual liability and benefits

Except as otherwise set out in this document, any agreement, covenant, representation or warranty under this document by two or more persons binds them jointly and each of them individually, and any benefit in favour of two or more persons is for the benefit of them jointly and each of them individually.

21.7 No fetter

Nothing in this document will be construed as requiring the Minister to do anything that would cause it to be in breach of any of its obligations at law, and without limitation, nothing will be construed as limiting or fettering in any way the exercise of any statutory discretion or duty.

21.8 Representations and warranties

The Parties represent and warrant that they have power to enter into this document and comply with their obligations under this document and that entry into this document will not result in the breach of any law.

21.9 Severability

If a clause or part of a clause of this document can be read in a way that makes it illegal, unenforceable or invalid, but can also be read in a way that makes it legal, enforceable and valid, it must be read in the latter way. If any clause or part of a clause is illegal, unenforceable or invalid, that clause or part is to be treated as removed from this document, but the rest of this document is not affected.

21.10 Waiver

The fact that a Party fails to do, or delays in doing, something the Party is entitled to do under this document, does not amount to a waiver of any obligation of, or breach of obligation by, another Party. A waiver by a Party is only effective if it is in writing. A written waiver by a Party is only effective in relation to the particular obligation or breach in respect of which it is given. It is not to be taken as an implied waiver of any other obligation or breach or as an implied waiver of that obligation or breach in relation to any other occasion.

21.11 Effect of Schedules

The Parties agree to comply with any terms contained in Schedules to this document as if those terms were included in the operative part of the document.

21.12 Relationship of parties

This document is not intended to create a partnership, joint venture or agency relationship between the Parties.

21.13 Further steps

Each Party must promptly do whatever any other Party reasonably requires of it to give effect to this document and to perform its obligations under it.

21.14 Counterparts

This document may consist of a number of counterparts and, if so, the counterparts taken together constitute one document.

21.15 Rights cumulative

Except as expressly stated otherwise in this document, the rights of a Party under this document are cumulative and are in addition to any other rights of that Party.

21.16 Liability

An obligation of two or more persons binds them separately and together.

21.17 Interpretation

Unless expressed to the contrary, in this document:

- (a) words in the singular include the plural and vice versa;
- (b) any gender includes the other genders;
- (c) if a word or phrase is defined its other grammatical forms have corresponding meanings;
- (d) 'includes' means includes without limitation;
- (e) a reference in this document to a Business Day means a day other than a Saturday or Sunday on which banks are open for business generally in Sydney;

- (f) if the day on or by which any act, must be done under this document is not a Business Day, the act must be done on or by the next Business Day;
- (g) '\$' or 'dollars' is a reference to Australian currency all amounts payable under this document are payable in Australian dollars;
- (h) a reference in this document to any law, legislation or legislative provision includes any statutory modification, amendment or re enactment, and any subordinate legislation or regulations issued under that legislation or legislative provision;
- a reference in this document to any agreement, deed or document is to that agreement, deed or document as amended, novated, supplemented or replaced;
- (j) a reference to a clause, part, schedule or attachment is a reference to a clause, part, schedule or attachment of or to this document;
- (k) an expression importing a natural person includes any company, trust, partnership, joint venture, association, body corporate or governmental agency;
- (I) a reference to a Party to this document includes a reference to the servants, agents and contractors of the Party, and the Party's successors and assigns; and
- (m) any schedules, annexures and attachments form part of this document.

21.18 Headings

Headings do not affect the interpretation of this document.

21.19 Effect of execution

This document is not binding on any Party unless it or a counterpart has been duly executed by each Party.

Schedule 1

Requirements under the Act

The Parties acknowledge and agree that the table set out below provides for certain terms, conditions and procedures for the purpose of the document complying with the Act.

Table 1 – Requirements under section 93F of the Act (clause 7)

| Prov | ision of the Act | This document |
|---|--|---|
| Under section 93F(1), the Landowners and the Developer have: | | |
| (a) | sought a change to an environmental planning instrument. | (a) Yes |
| (b) | made, or proposes to make, a development application. | (b) Yes |
| (c) | entered into an agreement with, or is otherwise associated with, a person, to whom paragraph (a) or (b) applies. | (c) No |
| | eription of the land to which this ument applies- (Section 93F(3)(a)) | The whole of the Land described in Schedule 2. |
| Description of the development to which this document applies- (Section 93F(3)(b)(ii)) | | The Development as defined in clause 4. |
| The scope, timing and manner of delivery of Development Contributions required by this document - (Section 93F(3)(c)) | | See Schedule 3. |
| Applicability of Section 94 of the Act - (Section 93F(3)(d)) | | The application of section 94 is not excluded in respect of any part of the Development. |
| | icability of Section 94A of the Act - tion 93F(3)(d)) | The application of section 94A is not excluded in respect of any part of the Development. |
| | icability of Section 94EF of the Act - tion 93F(3)(d)) | The application of section 94EF is wholly excluded in respect of any part of the Development. |

| Provision of the Act | This document |
|---|---|
| Applicability of Section 93F(3)(e) of the Act | Any benefits under this Agreement for a Public Purpose may be taken into consideration in determining a Development Contribution under section 94 of the Act. |
| Mechanism for Dispute resolution - (Section 93F(3)(f)) | See clause 9. |
| Enforcement of this document - (Section 93F(3)(g)) | See clauses 8, 10, Schedule 3 (clauses 3.2(c), 3.3 and 3.4) and Schedule 4. |
| No obligation to grant consent or exercise functions - (Section 93F(9)) | See clause 21.7. |

Table 1 – Other Matters

| Provision of the Act | This document |
|--|--------------------|
| Registration of this document (Section 93H) | Yes – see clause 8 |
| Whether this document specifies that certain requirements must be complied with before a construction certificate is issued (Clause 25E(2)(g) of the Regulation) | Yes |
| Whether this document specifies that certain requirements must be complied with before an occupation certificate is issued (Clause 25E(2)(g) of the Regulation) | No |
| Whether this document specifies that certain requirements must be complied with before an subdivision certificate is issued (Clause 25E(2)(g) of the Regulation) | Yes |

Schedule 2

Land

| Item | Contribution Area | Lot and Deposited Plan | Folio Identifier | Landowner |
|------|------------------------|---|---------------------|--|
| 1 | Contribution Area 1 | Lot 16 in DP225350 | 16/225350 | The Trust Company Limited as |
| | | Lot 17 in DP225350 | 17/225350 | |
| | | Lot 18 in DP225350 | 18/225350 | custodian for |
| | | Lot 19 in DP225350 | 19/225350 | the Carter Street Trust |
| | | Lot 23 in DP225350 | 23/225350 | |
| | | Lot 24 in DP225350 | 24/225350 | |
| | | Lot 200 in DP1160458 (the EnergyAustralia Land) | 200/1160458 | |
| 2 | Contribution Area 2 | Part Lot 25 in DP225350 | 25/225350 | The Trust Company Limited as custodian for the Carter Street Trust |
| | | Part Lot 2 in DP234801 | 2/234801 | The Trust Company Limited as custodian for the IBC Trust |
| 3 | Contribution Area 3 | Part Lot 2 in DP234801 | 2/234801 | The Trust Company Limited as custodian for the IBC Trust |
| 4 | Contribution | Part Lot 2 in DP234801 | 2/234801 | The Trust |
| | Area 4 | Lot 27 in DP225350 | 27/225350 | Company Limited as |
| | | Lot 28 in DP225350 | 28/225350 | custodian for the IBC Trust |
| | | Lot 301 in DP541070 | 301/541070 | The Trust Company Limited as custodian for |

| Item | Contribution Area | Lot and Deposited Plan | Folio Identifier | Landowner |
|------|------------------------|-------------------------|---------------------|--|
| | | | | the Hill Road Trust |
| 5 | Contribution Area 5 | Part Lot 25 in DP225350 | 25/225350 | The Trust Company Limited as custodian for the Carter Street Trust |
| | | Lot 26 in DP225350 | 26/225350 | The Trust Company Limited as custodian for the IBC Trust |
| | | Part Lot 2 in DP 234801 | 2/234801 | |

Schedule 3

Development Contributions

1 Development Contributions

The Developer and the Landowners undertake to provide the Development Contributions in accordance with the table below:

| Column 1 | Column 2 | Column 3 |
|--|--|--|
| Development Contribution | Value | Timing |
| Contribution Amount: payment of monetary contributions by the Developer or the relevant Landowner to the Minister. | \$100 per square metre of Allowable GFA, indexed in accordance with clause 2.2 of this Schedule. | In accordance with clause 2.3(a) of this Schedule. |
| Dedication of Education Land: transfer of the Education Land to the Minister or the Minister's nominee. | | In accordance with clause 3.1(c) of this Schedule. |

2 Contribution Amount

2.1 Calculation of the value of the Contribution Amount

Each Contribution Amount will be an amount equal to the sum represented by "A" in the following formula:

 $A = N \times 100$ (indexed in accordance with **clause 2.2** of this Schedule).

Where "N" is the number of square metres of Allowable GFA comprised in the land within the relevant Contribution Area.

2.2 Indexation

The value of the Monetary Contributions amount set out in Column 2 of the Table included at **clause 1** of this Schedule is to be subject to annual indexation at the commencement of each financial year to reflect any increase in the CPI, from the date of this Agreement until paid.

2.3 Delivery of Contribution Amounts

- (a) Subject to **clause 2.3(b)** of this Schedule, the Developer or the relevant Landowner must pay to the Minister or the Minister's nominee the Contribution Amount for all of the Allowable GFA within a Contribution Area prior to:
 - (i) the issue of any Planning Approval in respect of any part of that Contribution Area; or
 - (ii) the issue of any Complying Development Certificate, Construction Certificate or Subdivision Certificate in respect of any part of the Land within that Contribution Area.

whichever is the earlier.

- (b) Notwithstanding **clause 2.3(a)** of this Schedule, neither the Developer nor the Landowners will be required to pay any Contribution Amount prior to the issue of any Planning Approval, Complying Development Certificate, Construction Certificate or Subdivision Certificate for:
 - alterations or additions to any existing approved development or use of any part of the Land, being a development or use having the benefit of a Planning Approval that is valid and operative as at the date of commencement of this document;
 - (ii) a modification under section 96 of the Act to any existing approved development or use of any part of the Land, being a development or use having the benefit of a Planning Approval that is valid and operative as at the date of commencement of this document;
 - (iii) fitout or alterations to fitout of any buildings existing on any part of the Land as at the date of commencement of this document, provided that such fitout is not for the purposes of residential use of such buildings;
 - (iv) subdivision of the Land for the purpose of creating a Contribution Area and any roads which are required to be created for the purpose of creating a Contribution Area as identified on the Plan;
 - (v) subdivision of Contribution Area 3 for the purpose of creating a separate title for the Education Land as identified on the Plan;
 - (vi) Remediation of the Education Land in accordance with the terms of this document; or
 - (vii) any other purpose, but only with the prior written consent of the Minister, which may be withheld by the Minister in his absolute discretion.
- (c) The Contribution Amounts are made for the purpose of this Schedule when cleared funds are deposited by means of electronic funds transfer into a bank account nominated by the Minister.
- (d) Either the Developer or the relevant Landowner is to give the Minister not less than 20 Business Days written notice of:

- (i) its intention to pay a Contribution Amount; and
- (ii) the amount proposed to be paid.
- (e) Neither the Developer nor the Landowners are required to pay a Contribution Amount under this document until the Minister, after having received a notice under **clause 2.3(d)**, has given to the Developer or relevant Landowner (as applicable) a tax invoice for the amount of the contribution that the Developer or the relevant Landowner intends to pay.
- (f) Neither the Developer nor the Landowners may request from the Secretary a Satisfactory Arrangements Certificate in respect of any part of the Land until such time as:
 - (i) the obligation to make a monetary contribution under this Schedule in respect of that part of the Land has been satisfied in accordance with the requirements in this document; and
 - (ii) the Minister or his Department has provided written confirmation that this obligation has been so satisfied.

3 Dedication of Education Land

3.1 Transfer obligations

- (a) The Third Landowner must dedicate, and (while the Third Landowner owns the Education Land) the Developer must procure dedication of, the Education Land to the Minister or the Minister's nominee for a consideration of \$1.
- (b) Dedication of the Education Land is made for the purposes of **clause**3.1(a) of this Schedule when the Minister is given a form of transfer, executed by the Third Landowner and in registrable form under the *Real Property Act 1900* (NSW), that is effective to transfer title to the Education Land to Minister or the Minister's nominee when registered.
- (c) The Third Landowner must provide to the Minister, and the Developer must procure provision of, the form of transfer referred to in clause
 3.1(b) of this Schedule:
 - (i) subject to **clause 3.1(c)(ii)** of this Schedule, within 30 Business Days of 9 September 2026; or
 - (ii) where the leases that are registered on the Education Land as at the date of this document expire, are surrendered or are terminated (**Termination Date**) on a date earlier than 9 September 2025, within 12 months of the Termination Date.
- (d) To allow for the registration of the form of transfer referred to in clause 3.1(b) of this Schedule, the Third Landowner must do, and the Developer must procure, the following:

- cause the certificate(s) of title for the Education Land to be produced to the LPI or provide a direction allowing the certificate(s) of title to be used for that purpose;
- (ii) give to the Minister an irrevocable undertaking to deliver to the Minister or the Minister's nominee the certificate(s) of title if the certificate(s) are released to the Landowner by the LPI; and
- (iii) take any other necessary action to give effect to the transfer of the title of the Education Land to the Minister or the Minister's nominee free of all encumbrances (including any leases, mortgages, easements, covenants and voluntary planning agreements, but excluding statutory easements) and affectations (including any charge or liability for rates, taxes and charges) other than service easements or such other encumbrances as agreed with the Minister,

and the Minster agrees to execute, or procure execution by the Minister's nominee of, the form of transfer referred to in **clause 3.1(b)** of this Schedule and all other documents necessary to give effect to the transfer of title of the Education Land to the Minister or the Minister's nominee promptly upon provision of the form of transfer to the Minister under **clause 3.1(b)** of this Schedule, and to provide any other information within the Minister's knowledge which is reasonably required by the Third Landowner to enable the provision of servicing of the Education Land referred to in **clause 3.1(f)(ii)** of this Schedule subject to the Third Landowner's compliance with the Remediation obligations under **clause 3.2** of this Schedule.

- (e) Where the Education Land is dedicated to the Minister's nominee, the Third Landowner or the Developer must give the Secretary copies of the title search(es) for the Education Land showing the Minister's nominee as the registered proprietor of the Education Land within 10 Business Days of the relevant transfer being registered.
- (f) The Third Landowner must:
 - (i) pay all rates and Taxes owing in respect of the Education Land up to and including the date that the Third Landowner delivers the form of transfer and certificate(s) of title for the Education Land pursuant to clause 3.1(b) and clause 3.1(d) of this Schedule, after which time the Minister or the Minister's nominee will be responsible for any rates and Taxes in relation to the Education Land; and
 - (ii) prior to delivery of the form of transfer and certificate(s) of title for the Education Land pursuant to clause 3.1(b) and clause 3.1(d) of this Schedule and, at its cost, ensure and provide evidence to the Minister that the Education Land is serviced by roads, sewer, water, electricity and communications of a size and capacity that is suitable for the operation of a primary school including for public

sports and playing fields on the Education Land, subject to the requirement or permission of any servicing agency.

3.2 Remediation obligations

- (a) The Third Landowner must, prior to providing the form of transfer referred to in **clause 3.1(b)** of this Schedule, undertake the Remediation Works referred to in **clause 3.2(b)** of this Schedule and provide to the Minister a Site Audit Statement which certifies that the Education Land is suitable for the proposed uses of a primary school and public sports and playing fields, either unconditionally or subject to conditions which are acceptable to the Minister acting reasonably.
- (b) In this clause, Remediation Works means the Remediation of any Contamination over, on, under, above or migrating through or from the Education Land, including by:
 - (i) undertaking Remediation in accordance with any remedial action plan or voluntary management proposal relating to the Contamination or Remediation of the Education Land;
 - (ii) undertaking Remediation in accordance with the conditions of all approvals and/or licences required to be obtained from any Authority in order to complete the Remediation of the Education Land in accordance with the remedial action plan or any voluntary management proposal relating to the Contamination or Remediation of the Education Land;
 - (iii) engaging a Site Auditor to oversee the Remediation of the Education Land;
 - (iv) obtaining and complying with the conditions of any approvals and/or licences required to be obtained under any Environmental Law; and
 - (v) complying with the requirements, notices and directions issued or administered by any Authority in respect of Contamination of the Education Land.
- (c) At the time of providing to the Minister the Site Audit Statement referred to in **clause 3.2(a)** of this Schedule, the Third Landowner must provide security by way of a Bank Guarantee in the amount of \$1 million, to be retained by the Minister for a period of 12 months (**Retention Period**).
- (d) If at any time during the Retention Period, the Minister has any reasonable cause to believe that, notwithstanding the provision of any Site Audit Statement under clause 3.2(a) of this Schedule, Remediation of the Education Land was not undertaken to the standard required for the use specified under clause 3.2(a) of this Schedule, then the Minister may, at the Minister's absolute discretion, call upon the Bank Guarantee up to the amount necessary to investigate the environmental condition of the land and undertake any further Remediation for the purpose of Remediating the Education Land to that standard.

(e) At the end of the Retention Period, the Minister must return to the Third Landowner the whole or, subject to clause 3.2(d) of this Schedule, any remaining amount of, the security provided under clause 3.2(c) of this Schedule.

3.3 Compulsory acquisition

- (a) If the Developer and the Third Landowner do not comply with clause 3.1 of this Schedule, the Developer and the Third Landowner consent to the Minister or the Minister's nominee, at the appropriate time, compulsorily acquiring the Education Land in accordance with the Just Terms Act for an amount of \$1.
- (b) The Developer and the Third Landowner, and the Minister or the Minister's nominee, agree that in relation to the acquisition of the Education Land:
 - (i) **clause 3.3(a)** of this Schedule is taken to be an agreement between the Developer and the Third Landowner, and the Minister or the Minister's nominee, for the purpose of section 30 of the Just Terms Act; and
 - (ii) in clause 3.3(a) of this Schedule, the Developer and the Third Landowner, and the Minister or the Minister's nominee, acknowledge that they have agreed on all relevant matters concerning the compulsory acquisition and the compensation to be paid for the acquisition.
- (c) The Third Landowner and the Developer must ensure that the Education Land transferred under this **clause 3.3** is free of all encumbrances and affectations (including any charge or liability for rates, Taxes and charges) on the date that the Developer and the Third Landowner are liable to transfer the Education Land to the Minister or the Minister's nominee in accordance with this clause.
- (d) The Developer and the Third Landowner indemnify and agree to keep indemnified the Minister or the Minister's nominee against all claims made against the Minister or the Minister's nominee if the Minister or the Minister's nominee must pay compensation under Part 3 of the Just Terms Act to any person other than the Third Landowner.
- (e) The Developer and the Third Landowner must pay the Minister or the Minister's nominee, promptly on demand, an amount equivalent to all costs, charges and expenses incurred by the Minister or the Minister's nominee in acquiring the whole or any part of the Education Land and any other land required in connection with the Education Land as required by this clause.

3.4 Additional security

(a) The Minister reserves the right to call for additional security in relation to the transfer of the Education Land in accordance with this clause, but only where:

- the Developer or the Third Landowner provides written notification to the Minister prior to the transfer of the Education Land that it is proposed to transfer the Education Land subject to any of the encumbrances referred to in clause 3.1(d)(iii) of this Schedule (Encumbrances); or
- (ii) the Developer and the Third Landowner transfer the Education Land subject to Encumbrances.
- (b) Without limitation, the Minister may require the Developer or the Third Landowner to provide an additional Bank Guarantee, insurance bond or other agreed form of security in order to secure the costs of removing any Encumbrances noted on the title(s) to the Education Land. Clause 1 of Schedule 4 will apply to any Bank Guarantee provided under this clause.

3.5 Reimbursement of the Minister or Minister's nominee's costs

The Developer and the Third Landowner agree to reimburse the Minister or the Minister's nominee, promptly on demand, an amount equivalent to the Minister or the Minister's nominee's reasonable costs incurred in compulsorily acquiring the Education Land under **clause 3.3**. Where either the Developer or the Third Landowner has not complied, in whole or in part, with **clause 3.1** of this Schedule, this amount may include any costs associated with the Remediation of the Education Land to facilitate the use of the Education Land as a primary school and for public sports and playing fields.

Schedule 4

Security

1 Bank Guarantee

1.1 Developer and Landowners to provide Bank Guarantees

- (a) In order to secure the payment of each Contribution Amount, Registration of this document on title and the removal of any encumbrances as referred to in clause 3 of Schedule 3, the Developer and the Landowners have agreed to provide security in the form of a Bank Guarantee.
- (b) A Bank Guarantee required to be provided under this document must:
 - (i) name the "Minister for Planning and Department of Planning and Environment ABN 38 755 709 681" as the relevant beneficiaries; and
 - (ii) not have an expiry date.

1.2 Bank Guarantee

- (a) Upon execution of this document, the Developer and the Landowners will provide security to the Minister in the form of one or more Bank Guarantees for a face value equivalent to \$200,000, being security in the amount of \$40,000 for each Contribution Area.
- (b) From the date of execution of this document until the date that this document is released and discharged with respect to one or more Contribution Areas, the Minister will be entitled to retain the Bank Guarantee provided under clause 1.2(a) of this Schedule for the relevant Contribution Area or Contribution Areas.
- (c) Where one or more Bank Guarantees are required to be provided to the Minister under clause 8.1(g), such Bank Guarantee or Bank Guarantees must be provided in the amount(s) required by the Minister under that clause. From the date that the Bank Guarantee or Bank Guarantees are provided under clause 8.1(g) until the date that this document is registered on the corresponding titles comprising the Land, the Minister will be entitled to retain the Bank Guarantee or Bank Guarantees to secure registration of this document over the corresponding Lots comprising the Land.

1.3 Claims under the Bank Guarantee

(a) The Minister may:

- (i) call upon any Bank Guarantee provided under this document where the Developer and the Landowners have failed to pay a Contribution Amount in respect of the relevant part of the Land to which the Bank Guarantee relates on or before the due date for payment under this document, or are otherwise in default of their obligations under this document in respect of the relevant part of the Land to which the Bank Guarantee relates; and
- (ii) retain and apply such monies towards the costs and expenses incurred by the Minister in rectifying any default by the relevant Party under this document.
- (b) Prior to calling on a Bank Guarantee, the Minister must give the Developer or the relevant Landowner (as applicable) not less than 10 Business Days written notice.
- (c) If:
 - (i) the Minister calls upon any Bank Guarantee; and
 - (ii) applies all or part of such monies to towards the costs and expenses incurred by the Minister in rectifying any default or noncompliance by the Developer or a Landowner under this document; and
 - (iii) has notified the Developer or the relevant Landowner (as applicable) under **clause 1.3(b)** of this Schedule,

then the Developer or the relevant Landowner (as applicable) must provide to the Minister a replacement Bank Guarantee to ensure that at all times until the date that this document is released and discharged with respect to the whole of the Land, the Minister is in possession of:

- (A) a Bank Guarantee for a face value equivalent to \$200,000, being security in the amount of \$40,000 for each Contribution Area, subject to any increases resulting from the provision of additional security under clauses 13.1(b)(i) or 13.2(e)(ii); and
- (B) a Bank Guarantee or Bank Guarantees for any amount(s) required to be provided to the Minister pursuant to clause
 8.1(g) of this document.

1.4 Release of Bank Guarantee

If:

- (a) the Developer and the relevant Landowners have satisfied all of their obligations under this document secured by the relevant Bank Guarantee; and
- (b) the whole of the monies secured by the relevant Bank Guarantee have not been expended and the monies accounted for,

then the Minister will promptly return the relevant Bank Guarantee (less any costs, charges, duties and taxes payable) or the remainder of the monies

secured by the relevant Bank Guarantee (as the case may be) to the Developer and the relevant Landowners.

2 Charge over Land

2.1 Grant of charge

- (a) To support the caveat referred to in **clause 8.3** of the document, the Developer and the Landowners charge their rights, titles and interests in the Land in favour of the Minister to secure:
 - (i) the proper and timely performance of their obligations under this document; and
 - (ii) any damages that may be payable to the Minister in the event of any breach of this document by the Developer or any Landowner.
- (b) Upon execution of this document, the Developer is to, at its cost
 - (i) give the Minister a written instrument in registrable form under the Real Property Act 1900 (NSW) effective to register the charge on title to the Land;
 - (ii) obtain all necessary consents for the registration of the charge; and
 - (iii) produce the certificates of title for the Land to permit the registration of the charge.

2.2 Minister may caveat

For the purposes of **clause 8.3** of this document, the Developer and the Landowners consent to the Minister registering a caveat on the title of the Land to which the charge granted under **clause 2.1** of this Schedule applies.

2.3 Consent

The provisions in **clause 8.3(b)** of this document apply to the removal of the charge from the Land.

Schedule 5

Explanatory Note

Environmental Planning and Assessment Regulation 2000

(Clause 25E)

Explanatory Note

Draft Planning Agreement

Under section 93F of the Environmental Planning and Assessment Act 1979 (NSW)

1 Parties

The Minister for Planning of the Department of Planning and Environment, 23-33 Bridge Street, Sydney NSW 2000 (**Minister**)

Goodman Property Services (Aust) Pty Limited ABN 40 088 981 793 of Level 17, 60 Castlereagh Street, Sydney NSW 2000 (**Developer**)

The Trust Company Limited ABN 59 004 027 749 as custodian for the Carter Street Trust ABN 12 909 843 665(**First Landowner**)

The Trust Company Limited ABN 59 004 027 749 as custodian for the Hill Road Trust ABN 54 244 152 320 (**Second Landowner**)

The Trust Company Limited ABN 59 004 027 749 as custodian for the IBC Trust ABN 30 476 081 229 (**Third Landowner**)

Tallina Pty Ltd ACN 090 716 895 as trustee for the Carter Street Trust ABN 12 909 843 665

Tallina Pty Ltd ACN 090 716 895 as trustee for the Hill Road Trust ABN ABN 54 244 152 320

Tallina Pty Ltd ACN 090 716 895 as trustee for the IBC Trust ABN 30 476 081 229

2 Description of subject land

Lots 16 to 19 and 23 to 28 in DP225350, Lot 200 in DP1160458, Lot 2 in DP234801 and Lot 301 in DP541070 (**Land**).

3 Description of proposed change to environmental planning instrument

A new State Environmental Planning Policy (**Proposed SEPP**) and associated Development Control Plan is proposed. The Proposed SEPP will amend the *Auburn Local Environmental Plan 2010* (**ALEP**) to rezone the Land and vary the development controls for maximum allowable building heights and floor space ratios.

4 Summary of objectives, nature and effect of the draft planning agreement

4.1 Objectives of Planning Agreement

The objective of the Planning Agreement is to facilitate the development and improvement of public facilities within the Carter Street Priority Precinct and specifically to allow for:

- the provision of a monetary contribution to the Minister at the rate of \$100 per square metre of Allowable GFA permitted under the ALEP (subject to CPI indexation); and
- (b) the dedication of the Education Land for the purpose of a primary school operated for the benefit of the public by the Department of Education & Communities and for the purpose of public sports and playing fields.

4.2 Nature and effect of Planning Agreement

The Planning Agreement constitutes a planning agreement under section 93F of the Act. The Planning Agreement will be registered on the title to the Land.

5 Assessment of the merits of the draft planning agreement

5.1 The planning purposes served by the Planning Agreement

The Planning Agreement:

- (a) promotes the orderly and economic use and development of the Land to which the Planning Agreement applies;
- (b) facilitates the provision of land for public purposes in connection with the Planning Proposal and the Development;
- (c) facilitates the provision of community services and facilities; and

 promotes increased opportunity for the public in environmental planning and assessment.

5.2 How the Planning Agreement promotes the public interest

The Planning Agreement will promote the public interest by promoting the objects of the Act as set out in sections 5(a)(ii), (iv), (v) and 5(c) of the Act.

5.3 For Planning Authorities:

Development Corporations – How the Planning Agreement promotes its statutory responsibilities

Not Applicable

Other Public Authorities – How the Planning Agreement promotes the objects (if any) of the Act under which it is constituted

Not Applicable

Councils – How the Planning Agreement promotes the elements of the Council's Charter

Not Applicable

All Planning Authorities – Whether the Planning Agreement conforms with the Authority's Capital Works Program

Not applicable

5.4 Certain requirements of the Planning Agreement to be complied with before issue of Planning Approval or issue of a Complying Development Certificate, Construction Certificate or Subdivision Certificate

The Contribution Amount is to be paid in stages.

Prior to the issue of any Planning Approval in relation to any part of the Land in a Contribution Area or prior to the issue of any Complying Development Certificate, Construction Certificate or Subdivision Certificate for any part of the Land within that Contribution Area, whichever is the earlier, the Developer or the relevant Landowners must pay to the Minister the Contribution Amount for the whole of the Land within that Contribution Area. However the Developer or the relevant Landowners will not be required to pay any Contribution Amount prior to the issue of any Planning Approval, Complying Development Certificate, Construction Certificate or Subdivision Certificate for:

- (a) alterations or additions to any existing approved development or use of any part of the Land, being a development or use having the benefit of a Planning Approval that is valid and operative as at the date of commencement of the Planning Agreement;
- (b) a modification under section 96 of the Act to any existing approved development or use of any part of the Land, being a development or use having the benefit of a Planning Approval that is valid and operative as at the date of commencement of the Planning Agreement;

- (c) fitout or alterations to fitout of any buildings existing on any part of the Land as at the date of commencement of the Planning Agreement, provided that such fitout is not for the purposes of residential use of such buildings;
- (d) subdivision of the Land for the purpose of creating each Contribution
 Area and any roads which are required to be created for the purpose of creating each Contribution Area as identified in the Plan;
- (e) subdivision of Contribution Area 3 for the purpose of creating a separate title for the Education Land as identified in the Plan; or
- (f) remediation of the Education Land in accordance with the terms of the Planning Agreement.

The Third Landowner must provide to the Minister and the Developer must procure provision of, the form of transfer for the Education Land in accordance with the following timing:

- (i) within 30 Business Days of 9 September 2026; or
- (ii) where the leases that are registered on the Education Land as at the date of the Planning Agreement expire or are terminated (**Termination Date**) on a date earlier than 30 Business Days after 9 September 2026, within 12 months after the Termination Date.

The provision of the Development Contributions is secured through the requirement for a Bank Guarantee in respect of the Minister's enforcement costs, registration of a caveat supported by a charge over the Land until the Planning Agreement is registered on the title to the Land, the ability for the Minister to call for Bank Guarantees up to \$1 million per Lot comprising the Land where the Planning Agreement is not registered on all Lots comprising the Land within 90 Business Days (in addition to the charge and the caveat) and then by registration of the Planning Agreement on the title to the Land. The Minister also has the ability to compulsorily acquire the Education Land in accordance with the Land Acquisition (Just Terms Compensation) Act 1991 (NSW) if the Education Land is not dedicated in accordance with the Planning Agreement.

Execution

| Executed as a deed. |
|--|
| Minister |
| Signed sealed and delivered) by The Minister for Planning in the) presence of:) |
| Signature of Witness Signature of Minister |
| Name of Witness Name of Minister |
| Developer |
| Signed sealed and delivered for and on behalf of Goodman Property Services (Aust) Pty Limited by its attorney under power of attorney BookNowho states that no notice of revocation of the power of attorney has been received in the presence of: |
| Signature of witness |
| Name of witness (BLOCK LETTERS) |
| Address of witness |

13798319/4 Planning Agreement

First Landowner

| Signed sealed and delivered for and on behalf of The Trust Company Limited as custodian for the Carter Street Trust by its attorney under power of attorney BookNowho states that no notice of revocation of the power of attorney has been received in the presence of: | |
|--|--------------------------|
| Signature of witness | Attorney |
| Signature of withess | Allomey |
| Name of witness (BLOCK LETTERS) | Name of Attorney (print) |
| | |
| Address of witness | |
| Second Landowner | |
| Signed sealed and delivered for and on behalf of The Trust Company Limited as | |
| custodian for the Hill Road Trust by its attorney under power of attorney | |
| BookNowho states that no notice of revocation of the power of attorney has been received in the presence of: | |
| | |
| Signature of witness | Attorney |
| Name of witness (BLOCK LETTERS) | Name of Attorney (print) |
| | |
| Address of witness | |

13798319/4 Planning Agreement

Third Landowner

Signed sealed and delivered for and on behalf of The Trust Company Limited as custodian for the IBC Trust by its attorney under power of attorney Book.....No...who states that no notice of revocation of the power of attorney has been received in the presence of:

| power of attorney has been received in the presence of: | |
|--|--------------------------------------|
| | |
| Signature of witness | Attorney |
| Name of witness (BLOCK LETTERS) | Name of Attorney (print) Attorney |
| Address of witness | |
| Trustees | |
| Signed sealed and delivered for and on behalf of Tallina Pty Ltd as trustee for the Carter Street Trust by its attorney under power of attorney BookNowho states that no notice of revocation of the power of attorney has been received in the presence | |
| of: | |
| Signature of witness | Attorney |
| Name of witness (BLOCK LETTERS) | Name of Attorney (print) |

Address of witness

| Signed sealed and delivered for and on behalf of Tallina Pty Ltd as trustee for the Hill Road Trust by its attorney under power of attorney BookNowho states that no notice of revocation of the power of attorney has been received in the presence of: | |
|---|--------------------------------------|
| | • |
| Signature of witness | Attorney |
| Name of witness (BLOCK LETTERS) | Name of Attorney (print) |
| Address of witness | |
| Signed sealed and delivered for and on behalf of Tallina Pty Ltd as trustee for the IBC Trust by its attorney under power of attorney BookNowho states that no notice of revocation of the power of attorney has been received in the presence of: | |
| | |
| Signature of witness | Attorney |
| Name of witness (BLOCK LETTERS) | Name of Attorney (print) Attorney |
| Address of witness | |

Annexure A

Novation Deed



The Minister for Planning

Goodman Property Services (Aust) Pty Limited

The Trust Company Limited as custodian for the Carter Street Trust

The Trust Company Limited as custodian for the Hill Road Trust

The Trust Company Limited as custodian for the IBC Trust

Tallina Pty Ltd as trustee for the Carter Street Trust

Tallina Pty Ltd as trustee for the Hill Road Trust

Tallina Pty Ltd as trustee for the IBC Trust

[Insert New Party]

Deed of Novation for Planning Agreement

Contents

| 1 | Definitions | 2 |
|----|---|----|
| 2 | Novation | 2 |
| | 2.1 Original Agreement | 2 |
| | 2.2 Reference in Original Agreement | 3 |
| | 2.3 Address for notices | 3 |
| 3 | Affirmation of the Planning Agreement | 3 |
| 4 | Indemnities | 3 |
| 5 | Warranties and representations | 4 |
| | 5.1 Warranties | 4 |
| | 5.2 Survival of warranties | 4 |
| 6 | Limitation of liability for the Carter Street Trust | 4 |
| | 6.1 Definitions | 4 |
| | 6.2 Limitation of the First Landowner's Liability | 5 |
| | 6.3 Limitation of Trustee's Liability | 6 |
| 7 | Limitation of liability for the Hill Road Trust | 7 |
| | 7.1 Definitions | 7 |
| | 7.2 Limitation of the Second Landowner's Liability | 8 |
| | 7.3 Limitation of Trustee's Liability | 9 |
| 8 | Limitation of liability for the IBC Trust | 10 |
| | 8.1 Definitions | 10 |
| | 8.2 Limitation of the Third Landowner's Liability | 11 |
| | 8.3 Limitation of Trustee's Liability | 12 |
| 9 | Trustee's representations and warranties | 13 |
| | 9.1 Definitions | 13 |
| | 9.2 The Trustee's representations and warranties | 14 |
| | 9.3 Trustee's additional obligations | 16 |
| | 9.4 The Landowners' representations and warranties | 17 |
| | 9.5 The Landowners' additional obligations | 17 |
| | 9.6 Non-merger | 18 |
| 10 | GST | 18 |
| | 10.1 Construction | 18 |
| | 10.2 Consideration GST exclusive | 18 |
| | 10.3 Payment of GST | 19 |
| | 10.4 Timing of GST payment | 19 |
| | 10.5 Tax invoice | 19 |
| | 10.6 Adjustment event | 19 |
| | 10.7 Reimbursements | 19 |
| | 10.8 Calculations based on other amounts | 19 |
| | 10.9 No merger | 20 |

| 11 | Stamp duty and costs | 20 |
|-----|--------------------------------|----|
| 12 | Further acts | 20 |
| 13 | Amendment | 20 |
| 14 | Governing law and jurisdiction | 20 |
| 15 | Counterparts | 20 |
| 16 | General | 21 |
| | 16.1 Construction | 21 |
| | 16.2 Headings | 21 |
| | 16.3 Effect of execution | 21 |
| Exe | cution | 22 |



Date

Parties

The Minister for Planning ABN 38 755 709 681 of Level 15, 52 Martin Place, Sydney NSW 2000 (**Minister**)

Goodman Property Services (Aust) Pty Limited ABN 40 088 981 793 of Level 17, 60 Castlereagh Street, Sydney NSW 2000 (**Developer**)

The Trust Company Limited ABN 59 004 027 749 as custodian for the Carter Street Trust ABN 12 909 843 665 (**First Landowner**)

The Trust Company Limited ABN 59 004 027 749 as custodian for the Hill Road Trust ABN 54 244 152 320 (**Second Landowner**)

The Trust Company Limited ABN 59 004 027 749 as custodian for the IBC Trust ABN 30 476 081 229 (**Third Landowner**)

Tallina Pty Ltd ACN 090 716 895 as trustee for the Carter Street Trust ABN 12 909 843 665 (**First Trustee**)

Tallina Pty Ltd ACN 090 716 895 as trustee for the Hill Road Trust ABN 54 244 152 320 (**Second Trustee**)

Tallina Pty Ltd ACN 090 716 895 as trustee for the IBC Trust ABN 30 476 081 229 (**Third Trustee**)

[Insert name of party] ACN [Insert ACN] of [Insert address] (New Party)

Background

- A The Minister, the Developer, the First Landowner, the Second Landowner, the Third Landowner, the First Trustee, the Second Trustee and the Third Trustee are parties to the Planning Agreement.
- B The Planning Agreement relates to the whole of the Land.
- The [insert relevant transferor party/parties] have entered into the Dealing with the New Party in relation to the Transfer Land.
- D The Developer and [insert relevant transferor Land Owner(s) and Trustee(s)] wish to novate the Required Obligations, being their obligations under the Planning Agreement which relate to the Transfer Land, to the New Party.

Agreed terms

1 Definitions

In this document these terms have the following meanings:

Dealing [Insert description of dealing e.g. the sale of the Transfer

Land to the New Party].

Effective Date [Insert date]

GST Has the same meaning as in the GST Law.

GST Law Has the meaning given to that term in *A New Tax System*

(Goods and Services Tax) Act 1999 (Cth) and any other

Act or regulation relating to the imposition or

administration of the GST.

Land Has the meaning given to that term in the Planning

Agreement.

Planning The planning agreement dated [Insert date] and made between the Minister, the Developer, the First Landowner,

the Second Landowner, the Third Landowner, the First Trustee, the Second Trustee and the Third Trustee.

Required The [insert relevant transferor party's /parties'] obligations

Obligations under the Planning Agreement which relate to the Transfer Land, being: [itemise as required for clarity,

including substitution of Bank Guarantees prior to/simultaneously with existing Bank Guarantees, registration of VPA, provision of Contributions and any

arrangements in relation to caveat, if applicable]

Transfer Land [Insert description of land e.g. Contribution Area 1 (as that

term is defined in the Planning Agreement), being that

part of the Land subject to the Dealing.

2 Novation

2.1 Original Agreement

Subject to clause 2.2 and with effect from the Effective Date:

(a) the New Party is substituted for the Developer and the [insert relevant transferor Land Owner(s) and Trustee(s)] as a party to the Planning Agreement [Note: this clause 2.1(a) is to be deleted unless all of the transferors' obligations under the VPA are to be novated in relation to the Transfer Land e.g. if the whole of the Transfer Land is being transferred to the New Party];

- (b) the New Party will be required to perform the Required Obligations in accordance with the Planning Agreement; and
- (c) the Developer and the [insert relevant transferor Land Owner(s) and Trustee(s)] are released and discharged from the Required Obligations, and from all claims (whether for costs, damages, fees, expenses or otherwise) relating to the Required Obligations.

2.2 Reference in Original Agreement

All references to the Developer and the [insert relevant transferor Land Owner(s) and Trustees] in the Planning Agreement which relate to the Required Obligations are to be construed as references to the New Party.

2.3 Address for notices

The Minister must address all notices and communications to be given or made by it and which relate to the Required Obligations, to the New Party under the Planning Agreement to the following address:

New Party:

Address: [Insert address]

Fax: [Insert fax]

Contact Person: [Insert contact]
Email: [Insert email]

3 Affirmation of the Planning Agreement

The Planning Agreement will be read and construed subject to this document, and in all other respects the provisions of the Planning Agreement are ratified and confirmed, and, subject to the variation and novation contained in this document, the Planning Agreement will continue in full force and effect.

4 Indemnities

- (a) The New Party indemnifies the Developer and the [insert relevant transferor Land Owner(s) and Trustee(s)] on demand against all liabilities, claims, damages and loss which the Developer and [insert relevant transferor Land Owner(s) and Trustee(s)] suffer or incur in relation to the Required Obligations, including those which arise or relate to acts or omissions occurring on or after the Effective Date.
- (b) The New Party indemnifies the Minister in respect of any non-compliance, breach or default of the Planning Agreement in respect of the Transfer Land by the Developer or any relevant Landowner and [insert relevant transferor Land Owner(s) and Trustee(s)] on demand, including those which arise or relate to breaches, non-compliance or default in respect of the Transfer Land occurring before, on or after the Effective Date.

5 Warranties and representations

5.1 Warranties

Each party represents and warrants that, at the time of execution, and at the Effective Date:

- it has capacity unconditionally to execute, deliver and comply with its obligations under this document;
- it has taken all necessary action to authorise the unconditional execution and delivery of, and the compliance with, its obligations under this document;
- (c) this document is a valid and legally binding obligation and is enforceable against it by each other party in accordance with its terms; and
- (d) its unconditional execution and delivery of, and compliance with its obligations under, this document do not contravene:
 - (i) any law or directive from a government entity;
 - (ii) its constituent documents;
 - (iii) any agreement or instrument to which it is a party; or
 - (iv) any obligation of it to any other person.

5.2 Survival of warranties

The warranties and representations in clause 5.1 survive the execution of this document and the novation of the Planning Agreement.

6 Limitation of liability for the Carter Street Trust

The provisions of this clause 6 apply despite anything to the contrary in this document.

6.1 Definitions

In this clause 6:

- (a) **Assets** includes all assets, property and rights, real and personal, of any value whatsoever, of the Trust.
- (b) **Constitution** means the constitution of the Trust as amended from time to time.
- (c) **Custody Agreement** means the Carter Street Custody Agreement dated 11 February 2002.
- (d) Obligations means all obligations and liabilities of whatever kind undertaken or incurred by, or devolving upon, the First Landowner under or in respect of this document.
- (e) Tallina means Tallina Pty Ltd ACN 090 716 895.
- (f) Trust means the Carter Street Trust ABN 12 909 843 665.

- (g) **Trustee** means the entity from time to time acting in the Trustee's Capacity, which on the making of this document is Tallina.
- (h) **Trustee's Capacity** means the capacity in which the Trustee enters into this document, being as trustee of the Trust.

6.2 Limitation of the First Landowner's Liability

- (a) The First Landowner enters into this document as custodian for the Trust and in no other capacity.
- (b) The parties other than the First Landowner acknowledge that the Obligations are incurred by the First Landowner solely in its capacity as custodian of the Assets and that the First Landowner will cease to have any Obligation under this document if the First Landowner ceases for any reason to be owner of the Assets.
- (c) The First Landowner will not be liable to pay or satisfy any Obligations except to the extent to which it is indemnified or entitled to be indemnified:
 - (i) by the Trustee; or
 - (ii) out of the Assets in respect of any liability incurred by it.
- (d) The parties other than the First Landowner may enforce their rights against the First Landowner arising from non-performance of the Obligations only to the extent of the First Landowner indemnities referred to in clause 6.2(c).
- (e) Subject to clause 6.2(e)(i), if any party other than the First Landowner does not recover all money owing to it arising from non-performance of the Obligations it may not seek to recover the shortfall by:
 - (i) bringing proceedings against the First Landowner in its personal capacity; or
 - (ii) applying to have the First Landowner wound up or proving in the winding up of the First Landowner.
- (f) The provisions of this clause 6.2 do not apply to any obligation or liability of the First Landowner to the extent that it is not satisfied because, under the Custody Agreement or by operation of law, there is a reduction in the extent of the First Landowner's indemnification from the Trustee or out of the Assets as a result of the First Landowner's fraud, negligence or breach of duty.
- (g) Except in the case of and to the extent of fraud, negligence or breach of duty on the part of the First Landowner under its custody agreement with the Trustee, the parties other than the First Landowner waive their rights and release the First Landowner from any personal liability whatsoever, in respect of any Loss or damage which they may suffer as a result of any:
 - (i) breach by the First Landowner of any of its Obligations; or

(ii) non-performance by the First Landowner of the Obligations but only to the extent to which the liability cannot be paid or satisfied by the indemnities set out above in clause 6.2(c)in respect of any liability incurred by it.

The parties other than The First Landowner acknowledge that the whole of this document is subject to this clause.

- (h) The parties acknowledge that the Trustee is responsible under the Constitution for performing a variety of obligations relating to the Trust, including under this document. The parties agree that no act or omission of the First Landowner (including any related failure to satisfy any Obligations) will constitute fraud, negligence or breach of duty of the First Landowner for the purposes of clause 6.2(e)(i) to the extent to which the act or omission was caused or contributed to by any failure of the Trustee or any other person to fulfil its obligations relating to the Trust or by any other act or omission of the Trustee or any other person.
- (i) No attorney, agent or other person appointed in accordance with this document has authority to act on behalf of the First Landowner in a way which exposes the First Landowner to any personal liability and no act or omission of such a person will be considered fraud, negligence or breach of duty of the First Landowner for the purposes of clause 6.2(e)(i).

6.3 Limitation of Trustee's Liability

(a) Capacity

The Trustee's liability under this document is limited to the Trustee's Capacity and the Trustee is not liable in any other capacity.

(b) Limitation

Subject to clause 6.3(d), the liability of the Trustee in respect of any cause of action, Claim or Loss arising:

- (i) under or in connection with this document;
- (ii) in connection with any transaction, conduct or any other agreement contemplated by this document; or
- (iii) under or in connection with (to the extent permitted by law) any representation or undertaking given or to be given in connection with this document,

(each, a **Trust Claim**) and the right of each party other than the Trustee to recover from the Trustee any amount in respect of any (and all) Trust Claims is limited to a right to recover an amount not exceeding the amount which the Trustee is entitled and able to recover from the Assets (after taking account of the costs of exercising its right of indemnity or exoneration) and if, after exercise of those rights, any such amount remains outstanding, no further Trust Claim may be made against the Trustee personally in respect of that outstanding amount.

(c) Acknowledgment of limitations

The parties other than the Trustee agree and acknowledge that they must not, in respect of any Trust Claim:

- (i) subject to clause 6.3(d), bring proceedings against the Trustee in its personal capacity;
- (ii) seek to appoint an administrator or liquidator to the Trustee;
- (iii) commence the winding-up, dissolution or administration of the Trustee; or
- (iv) appoint a receiver, receiver and manager, administrative receiver or similar official to all or any of the assets of the Trustee,

except to the extent that the steps taken affect any Assets or the Trustee's right of recourse against, and indemnity from, the Assets and nothing else.

(d) Exception

If the Trustee acts negligently, fraudulently, with wilful misconduct or in breach of trust with a result that:

- (i) the Trustee's right of indemnity, exoneration or recoupment of the Assets; or
- (ii) the actual amount recoverable by the Trustee in exercise of those rights,

is reduced in whole or in part or does not exist, then to the extent that such right or the amount so recoverable is reduced or does not exist, the Trustee will be personally liable.

7 Limitation of liability for the Hill Road Trust

The provisions of this clause **7** apply despite anything to the contrary in this document.

7.1 Definitions

In this clause 7:

- (a) **Assets** includes all assets, property and rights, real and personal, of any value whatsoever, of the Trust.
- (b) **Constitution** means the constitution of the Trust as amended from time to time.
- (c) **Custody Agreement** means the Hill Road Trust Custody Agreement dated 30 October 2002.
- (d) Obligations means all obligations and liabilities of whatever kind undertaken or incurred by, or devolving upon, the Second Landowner under or in respect of this document.

Tallina means Tallina Pty Ltd ACN 090 716 895.

- (e) Trust means the Hill Road Trust ABN 54 244 152 320.
- (f) **Trustee** means the entity from time to time acting in the Trustee's Capacity which on the making of this document is Tallina.
- (g) **Trustee's Capacity** means the capacity in which the Trustee enters into this document, being as trustee of the Trust.

7.2 Limitation of the Second Landowner's Liability

- (a) The Second Landowner enters into this document as custodian for the Trust and in no other capacity.
- (b) The parties other than the Second Landowner acknowledge that the Obligations are incurred by the Second Landowner solely in its capacity as custodian of the Assets and that the Second Landowner will cease to have any Obligation under this document if the Second Landowner ceases for any reason to be owner of the Assets.
- (c) The Second Landowner will not be liable to pay or satisfy any
 Obligations except to the extent to which it is indemnified or entitled to be indemnified:
 - (i) by the Trustee; or
 - (ii) out of the Assets in respect of any liability incurred by it.
- (d) The parties other than the Second Landowner may enforce their rights against the Second Landowner arising from non-performance of the Obligations only to the extent of the Second Landowner indemnities referred to in clause 7.2(c).
- (e) Subject to clause 7.2(e)(i), if any party other than the Second Landowner does not recover all money owing to it arising from non-performance of the Obligations it may not seek to recover the shortfall by:
 - (i) bringing proceedings against the Second Landowner in its personal capacity; or
 - (ii) applying to have the Second Landowner wound up or proving in the winding up of the Second Landowner.
- (f) The provisions of this clause 7.2 do not apply to any obligation or liability of the Second Landowner to the extent that it is not satisfied because, under the Custody Agreement or by operation of law, there is a reduction in the extent of the Second Landowner's indemnification from the Trustee or out of the Assets as a result of the Second Landowner's fraud, negligence or breach of duty.
- (g) Except in the case of and to the extent of fraud, negligence or breach of duty on the part of the Second Landowner under its custody agreement with the Trustee, the parties other than the Second Landowner waive their rights and release the Second Landowner from any personal liability whatsoever, in respect of any Loss or damage which they may suffer as a result of any:

- (i) breach by the Second Landowner of any of its Obligations; or
- (ii) non-performance by the Second Landowner of the Obligations, but only to the extent to which the liability cannot be paid or satisfied by the indemnities set out above in clause 7.2(c) in respect of any liability incurred by it.

The parties other than The Second Landowner acknowledge that the whole of this document is subject to this clause.

- (h) The parties acknowledge that the Trustee is responsible under the Constitution for performing a variety of obligations relating to the Trust, including under this document. The parties agree that no act or omission of the Second Landowner (including any related failure to satisfy any Obligations) will constitute fraud, negligence or breach of duty of the Second Landowner for the purposes of clause 7.2(e)(i) to the extent to which the act or omission was caused or contributed to by any failure of the Trustee or any other person to fulfil its obligations relating to the Trust or by any other act or omission of the Trustee or any other person.
- (i) No attorney, agent or other person appointed in accordance with this document has authority to act on behalf of the Second Landowner in a way which exposes the Second Landowner to any personal liability and no act or omission of such a person will be considered fraud, negligence or breach of duty of the Second Landowner for the purposes of clause 7.2(e)(i).

7.3 Limitation of Trustee's Liability

(a) Capacity

The Trustee's liability under this document is limited to the Trustee's Capacity and the Trustee is not liable in any other capacity.

(b) Limitation

Subject to clause 7.3(d), the liability of the Trustee in respect of any cause of action, Claim or Loss arising:

- (i) under or in connection with this document;
- (ii) in connection with any transaction, conduct or any other agreement contemplated by this document; or
- (iii) under or in connection with (to the extent permitted by law) any representation or undertaking given or to be given in connection with this document,

(each, a **Trust Claim**) and the right of each party other than the Trustee to recover any amount from the Trustee in respect of any (and all) Trust Claims is limited to a right to recover an amount not exceeding the amount which the Trustee is entitled and able to recover from the Assets (after taking account of the costs of exercising its right of indemnity or exoneration) and if, after exercise of those rights, any such amount

remains outstanding, no further Trust Claim may be made against the Trustee personally in respect of that outstanding amount.

(c) Acknowledgment of limitations

The parties other than the Trustee agree and acknowledge that they must not, in respect of any Trust Claim:

- (i) subject to clause 7.3(d), bring proceedings against the Trustee in its personal capacity;
- (ii) seek to appoint an administrator or liquidator to the Trustee;
- (iii) commence the winding-up, dissolution or administration of the Trustee; or
- (iv) appoint a receiver, receiver and manager, administrative receiver or similar official to all or any of the assets of the Trustee,

except to the extent that the steps taken affect any Assets or the Trustee's right of recourse against, and indemnity from, the Assets and nothing else.

(d) Exception

If the Trustee acts negligently, fraudulently, with wilful misconduct or in breach of trust with a result that:

- (i) the Trustee's right of indemnity, exoneration or recoupment of the Assets; or
- (ii) the actual amount recoverable by the Trustee in exercise of those rights,

is reduced in whole or in part or does not exist, then to the extent that such right or the amount so recoverable is reduced or does not exist, the Trustee will be personally liable.

8 Limitation of liability for the IBC Trust

The provisions of this clause 8 apply despite anything to the contrary in this document.

8.1 Definitions

In this clause 8:

- (a) **Assets** includes all assets, property and rights, real and personal, of any value whatsoever, of the Trust.
- (b) **Constitution** means the constitution of the Trust as amended from time to time.
- (c) Custody Agreement means the IBC Trust Custody Agreement dated 11 May 2005.

- (d) **Obligations** means all obligations and liabilities of whatever kind undertaken or incurred by, or devolving upon, the Third Landowner under or in respect of this document.
- (e) **Tallina** means Tallina Pty Ltd ACN 090 716 895.
- (f) **Trust** means the IBC Trust ABN 30 476 081 229.
- (g) **Trustee** means the entity from time to time acting in the Trustee's Capacity which on the making of this document is Tallina.
- (h) **Trustee's Capacity** means the capacity in which the Trustee enters into this document, being as trustee of the Trust.

8.2 Limitation of the Third Landowner's Liability

- (a) The Third Landowner enters into this document as custodian for the Trust and in no other capacity.
- (b) The parties other than the Third Landowner acknowledge that the Obligations are incurred by the Third Landowner solely in its capacity as custodian of the Assets and that the Third Landowner will cease to have any Obligation under this document if the Third Landowner ceases for any reason to be owner of the Assets.
- (c) The Third Landowner will not be liable to pay or satisfy any Obligations except to the extent to which it is indemnified or entitled to be indemnified:
 - (i) by the Trustee; or
 - (ii) out of the Assets in respect of any liability incurred by it.
- (d) The parties other than the Third Landowner may enforce their rights against the Third Landowner arising from non-performance of the Obligations only to the extent of the Third Landowner indemnities referred to in clause 8.2(c).
- (e) Subject to clause 8.2(e)(i), if any party other than the Third Landowner does not recover all money owing to it arising from non-performance of the Obligations it may not seek to recover the shortfall by:
 - (i) bringing proceedings against the Third Landowner in its personal capacity; or
 - (ii) applying to have the Third Landowner wound up or proving in the winding up of the Third Landowner.
- (f) The provisions of this clause 8.2 do not apply to any obligation or liability of the Third Landowner to the extent that it is not satisfied because, under the Custody Agreement or by operation of law, there is a reduction in the extent of the Third Landowner's indemnification from the Trustee or out of the Assets as a result of the Third Landowner's fraud, negligence or breach of duty.

- (g) Except in the case of and to the extent of fraud, negligence or breach of duty on the part of the Third Landowner under its custody agreement with the Trustee, the parties other than the Third Landowner waive their rights and release the Third Landowner from any personal liability whatsoever, in respect of any Loss or damage which they may suffer as a result of any:
 - (i) breach by the Third Landowner of any of its Obligations; or
 - (ii) non-performance by the Third Landowner of the Obligations but only to the extent to which the liability cannot be paid or satisfied by the indemnities set out above in clause 8.2(c) in respect of any liability incurred by it.

The parties other than The Third Landowner acknowledge that the whole of this document is subject to this clause.

- (h) The parties acknowledge that the Trustee is responsible under the Constitution for performing a variety of obligations relating to the Trust, including under this document. The parties agree that no act or omission of the Third Landowner (including any related failure to satisfy any Obligations) will constitute fraud, negligence or breach of duty of the Third Landowner for the purposes of clause 8.2(e)(i) to the extent to which the act or omission was caused or contributed to by any failure of the Trustee or any other person to fulfil its obligations relating to the Trust or by any other act or omission of the Trustee or any other person.
- (i) No attorney, agent or other person appointed in accordance with this document has authority to act on behalf of the Third Landowner in a way which exposes the Third Landowner to any personal liability and no act or omission of such a person will be considered fraud, negligence or breach of duty of the Third Landowner for the purposes of clause 8.2(e)(i).

8.3 Limitation of Trustee's Liability

(a) Capacity

The Trustee's liability under this document is limited to the Trustee's Capacity and the Trustee is not liable in any other capacity.

(b) Limitation

Subject to clause 8.3(d), the liability of the Trustee in respect of any cause of action, Claim or Loss arising:

- (i) under or in connection with this document;
- (ii) in connection with any transaction, conduct or any other agreement contemplated by this document; or
- (iii) under or in connection with (to the extent permitted by law) any representation or undertaking given or to be given in connection with this document,

(each, a **Trust Claim**) and the right of each party other than the Trustee to recover any amount from the Trustee in respect of any (and all) Trust Claims is limited to a right to recover an amount not exceeding the amount which the Trustee is entitled and able to recover from the Assets (after taking account of the costs of exercising its right of indemnity or exoneration) and if, after exercise of those rights, any such amount remains outstanding, no further Trust Claim may be made against the Trustee personally in respect of the outstanding amount.

(c) Acknowledgment of limitations

The parties other than the Trustee agree and acknowledge that they must not, in respect of any Trust Claim:

- (i) subject to clause 8.3(d), bring proceedings against the Trustee in its personal capacity;
- (ii) seek to appoint an administrator or liquidator to the Trustee;
- (iii) commence the winding-up, dissolution or administration of the Trustee; or
- (iv) appoint a receiver, receiver and manager, administrative receiver or similar official to all or any of the assets of the Trustee,

except to the extent that the steps taken affect any Assets or the Trustee's right of recourse against, and indemnity from, the Assets and nothing else.

(d) Exception

If the Trustee acts negligently, fraudulently, with wilful misconduct or in breach of trust with a result that:

- (i) the Trustee's right of indemnity, exoneration or recoupment of the Assets; or
- (ii) the actual amount recoverable by the Trustee in exercise of those rights,

is reduced in whole or in part or does not exist, then to the extent that such right or the amount so recoverable is reduced or does not exist, the Trustee will be personally liable.

9 Trustee's representations and warranties

9.1 Definitions

In this clause 9:

- (a) **Assets** includes all assets, property and rights, real and personal, of any value whatsoever, of the Trusts.
- (b) **Beneficiaries** means the beneficiaries of the Trusts.
- (c) **Custody Agreements** means the following:

- (i) the IBC Trust Custody Agreement dated11 May 2005;
- (ii) the Hill Road Trust Custody Agreement dated 30 October 2002;
- (iii) the Carter Street Custody Agreement dated 11 February 2002.
- (d) Obligations means all obligations and liabilities of whatever kind undertaken or incurred by, or devolving upon, the First Landowner, the Second Landowner and the Third Landowner under or in respect of this document.
- (e) **Trusts** means the Carter Street Trust ABN 12 909 843 665, the Hill Road Trust ABN 54 244 152 320 and the IBC Trust 30 476 081 229.
- (f) **Trust Deeds** means:
 - (i) the IBC Trust Constitution dated 11 May 2005 between the Trustee and the relevant unit holders;
 - (ii) the Hill Road Trust Constitution dated 30 October 2002 between the Trustee and the relevant unit holders; and
 - (iii) the Carter Street Trust Constitution dated 11 February 2002 between the Trustee and the relevant unit holders.
- (g) **Trustee** means the entity from time to time acting in the Trustee's Capacity, which on the making of this document is Tallina.
- (h) **Trustee's Capacity** means the capacity in which the Trustee enters into this document, being as trustee of the Trust.
- (i) **Trustee Documents** means this document and any documents created under or in connection with this document.

9.2 The Trustee's representations and warranties

The Trustee represents and warrants to the Minister that:

- (a) powers of Trustee and custodian:
 - (i) the Trustee has power as trustee of the Trusts to execute the Trustee Documents; and
 - (ii) each Landowner has power as custodian of the relevant Trusts to execute the Trustee Documents;
- (b) **due performance:** in executing the Trustee Documents, the Trustee and its representatives have properly performed their obligations to the Beneficiaries;
- (c) **Trustee action taken:** all necessary action required by the Trust Deeds and the Custody Agreement to authorise the unconditional execution and delivery of, and compliance with the Trustee's obligations under, the Trustee Documents has been taken;
- (d) **custodian action taken:** all necessary action required by the Trust Deeds and the Custody Agreement to authorise the unconditional

- execution and delivery of, and compliance with each of the Landowner's obligations under, the Trustee Documents has been taken;
- (e) Trustee instructions the Trustee will give all necessary instructions to all Landowners to enable the Landowners to execute and deliver, and comply with all of their obligations under, the Trustee Documents;

(f) sole Trustee and custodian:

- (i) the Trustee is the only trustee of the Trusts; and
- (ii) the Landowners are the only custodians of the relevant Trusts;
- (g) **no removal of Trustee:** no action has been taken to remove the Trustee as trustee of the Trusts or to appoint an additional trustee of the Trusts;
- (h) **no removal of custodian:** no action has been taken to remove any or all of the Landowners as custodians of the relevant Trusts or to appoint any additional custodian of the relevant Trusts;

(i) Trustee's right of indemnity:

- (i) the Trustee has a right to be fully indemnified out of the Assets;
- (ii) the Trustee has not released or disposed of the Trustee's equitable lien over the Assets which secures that indemnity;
- (iii) the Trustee has not committed any breach of trust or done or omitted to do anything which has prejudiced or limited its rights of indemnity or equitable lien; and
- (iv) the Assets are sufficient to satisfy that indemnity;

(j) custodian's right of indemnity:

- (i) each Landowner has a right to be fully indemnified from the Assets and by the Trustee for all liabilities under the Trustee Documents;
- (ii) the Landowners have not released or disposed of the Landowners' equitable lien over the Assets which secures that indemnity;
- (iii) the Landowners have not committed any breach of the Custody Agreement or done or omitted to do anything which has prejudiced or limited their rights of indemnity or equitable lien; and
- (iv) the Assets are sufficient to satisfy that indemnity;

(k) no default:

- (i) the Trustee has not defaulted in the performance and observance of its obligations as trustee of the Trusts; and
- (ii) so far as the Trustee is aware, the Landowners have not defaulted in the performance and observance of their obligations as custodians of the Trusts or under the Custody Agreement so far as the Trustee is aware, the Landowners have not defaulted in the performance and observance of their obligations as custodians of the Trusts or under the Custody Agreement;

- (I) **Trusts or Custody Agreement not terminated:** no action has been taken or, so far as the Trustee is aware, is contemplated to terminate the Trusts or the Custody Agreement;
- (m) full disclosure: the Trustee has disclosed to the Minister full particulars of the Trusts, the Custody Agreement and any other trust or fiduciary relationship affecting the Assets and, without limitation, has given to the Minister copies of any instruments creating or evidencing the Trusts and the Custody Agreement;
- (n) **Trusts duly constituted:** the Trusts are duly constituted and are not void, voidable or otherwise unenforceable;
- (o) **custodian duly appointed:** each Landowner has been duly appointed as custodian of the relevant Trusts;
- (p) capacity of Trustee: the Trustee:
 - is duly incorporated in accordance with the laws of its place of incorporation as stated in this document, validly exists under those laws and has the capacity to sue and be sued in its own name, to own property and to act as trustee of the Trusts; and
 - (ii) is not insolvent.

9.3 Trustee's additional obligations

- (a) The Trustee must:
 - (i) ensure the Assets are not mixed with any other property;
 - (ii) comply with the Trustee's obligations as trustee of the Trusts;
 - (iii) not release, dispose of or otherwise prejudice the Trustee's right of indemnity against, and equitable lien over, the Assets and the Trustee's right of indemnity (if any) against the Beneficiaries;
 - (iv) at the Minister's request, exercise the Trustee's right of indemnity against, and equitable lien over, the Assets and the Trustee's right of indemnity (if any) against the Beneficiaries; and
 - (v) not, without obtaining the Minister's prior written approval:
 - (A) terminate any or all of the Trusts;
 - (B) resettle the Assets; or
 - (C) vary the terms of the Trust if to do so would adversely affect the Trustee's right of indemnity against, or equitable lien over, the Assets or the Trustee's right of indemnity (if any) against the Beneficiaries.
- (b) The Trustee covenants that it will not retire or be replaced as trustee of any or all of the Trusts unless:
 - (i) another entity has been appointed in its place for the relevant Trusts:

- the new entity agrees to be bound by the Trustee Documents, the Trust Deeds and the Custody Agreements (as applicable) as trustee for the relevant parcel of Land; and
- (iii) unless the new entity is a Developer-Related Entity, the Trustee obtains the Minister's prior written consent, which must not be unreasonably withheld, to the appointment of the new entity.

9.4 The Landowners' representations and warranties

Each Landowner represents and warrants to the Minister that:

- (a) **custodian's power:** the Landowner has power as custodian of the relevant Trusts to execute the Trustee Documents:
- (b) **due performance:** in executing the Trustee Documents, the Landowner and its representatives have properly performed their obligations to the Beneficiaries and Trustee:
- (c) custodian action taken: all necessary action required by the Trust Deeds and the Custody Agreement to authorise the unconditional execution and delivery of, and compliance with the Landowner's obligations under, the Trustee Documents has been taken;
- (d) **no removal of custodian:** so far as the Landowner is aware, no action has been taken to remove the Landowner as custodian of the relevant Trusts or to appoint an additional custodian of the relevant Trusts;
- (e) **custodian duly appointed:** each Landowner has been duly appointed as custodian of the relevant Trusts;
- (f) capacity of custodian: the Landowner:
 - is duly incorporated in accordance with the laws of its place of incorporation as stated in this document, validly exists under those laws and has the capacity to sue and be sued in its own name, to own property and to act as custodian of the Trusts; and
 - (ii) is not insolvent.

9.5 The Landowners' additional obligations

- (a) Each Landowner must:
 - (i) ensure the Assets are not mixed with any other property;
 - (ii) comply with the its obligations under the Custody Agreements as custodian of the relevant Trusts;
 - (iii) not release, dispose of or otherwise prejudice the Landowner's right of indemnity against, and equitable lien over, the Assets and the Landowner's right of indemnity (if any) against the Beneficiaries;

- (iv) at the Minister's request, exercise the Landowner's right of indemnity against, and equitable lien over, the Assets and the Landowner's right of indemnity (if any) against the Beneficiaries;
- (b) Each Landowner covenants that it will not retire or be replaced (and each Trustee covenants that it will not remove the Landowner or cause the Landowner to retire) as custodian of any or all of the Trusts unless:
 - (i) another entity has been appointed in its place for the relevant Trusts; and
 - (ii) the new entity agrees to be bound by the Trustee Documents, the Trust Deeds and the Custody Agreements (as applicable) as Landowner for the relevant parcel of Land.
- (c) No Landowner may take any action to amend the Trust Deeds or the Custody Agreements without the prior written consent of the Minister, acting reasonably.

9.6 Non-merger

This **clause 9** does not merge on the completion, rescission or other termination of this document or on the provision of any Development Contributions under this document.

10 GST

10.1 Construction

In this clause 10:

- unless there is a contrary indication, words and expressions which are not defined in this document but which have a defined meaning in the GST Law have the same meaning as in the GST Law; and
- (b) references to GST payable and input tax credit entitlements include:
 - (i) notional GST payable by, and notional input tax credit entitlements of the Commonwealth, a State or a Territory (including a government, government body, authority, agency or instrumentality of the Commonwealth, a State or a Territory); and
 - (ii) GST payable by, and the input tax credit entitlements of, the representative member of a GST group of which the entity is a member.

10.2 Consideration GST exclusive

Unless otherwise expressly stated, all consideration, whether monetary or non-monetary, payable or to be provided under or in connection with this document is exclusive of GST (**GST-exclusive consideration**).

10.3 Payment of GST

If GST is payable on any supply made by:

- (a) a party; or
- (b) an entity that is taken under the GST Law to make the supply by reason of the capacity in which a party acts,

(**Supplier**) under or in connection with this document, the recipient of the supply, or the party providing the consideration for the supply, must pay to the Supplier an amount equal to the GST payable on the supply.

10.4 Timing of GST payment

The amount referred to in clause 10.3 must be paid in addition to and at the same time and in the same manner (without any set-off or deduction) that the GST-exclusive consideration for the supply is payable or to be provided.

10.5 Tax invoice

The Supplier must deliver a tax invoice or an adjustment note to the recipient of a taxable supply before the Supplier is entitled to payment of an amount under clause 10.3.

10.6 Adjustment event

If an adjustment event arises in respect of a supply made by a Supplier under or in connection with this document, any amount that is payable under clause 10.3 will be calculated or recalculated to reflect the adjustment event and a payment will be made by the recipient to the Supplier or by the Supplier to the recipient as the case requires.

10.7 Reimbursements

- (a) Where a party is required under or in connection with this document to pay for, reimburse or contribute to any expense, loss, liability or outgoing suffered or incurred by another party or indemnify another party in relation to such an expense, loss, liability or outgoing (**Reimbursable Expense**), the amount required to be paid, reimbursed or contributed by the first party will be reduced by the amount of any input tax credits to which the other party is entitled in respect of the Reimbursable Expense.
- (b) This clause 10.7 does not limit the application of clause 10.3, if appropriate, to the Reimbursable Expense as reduced in accordance with clause 10.7(a).

10.8 Calculations based on other amounts

If an amount of consideration payable or to be provided under or in connection with this document is to be calculated by reference to:

(a) any expense, loss, liability or outgoing suffered or incurred by another person (Cost), that reference will be to the amount of that Cost excluding the amount of any input tax credit entitlement of that person relating to the Cost suffered or incurred; and (b) any price, value, sales, proceeds, revenue or similar amount (**Revenue**), that reference will be to that Revenue determined by deducting from it an amount equal to the GST payable on the supply for which it is consideration.

10.9 No merger

This clause 10 does not merge on the completion, rescission or other termination of this document or on the transfer of any property supplied under this document.

11 Stamp duty and costs

The Developer, [insert relevant transferor Land Owner(s) and Trustee(s)] and the New Party are jointly and severally responsible for the Minister's legal costs incidental to the negotiation, preparation and execution of this document.

12 Further acts

- (a) Each party will take all steps, execute all deeds and do everything reasonably required by any other party to give effect to any of the actions contemplated by this document.
- (b) This document binds each party which signs it even if other parties do not, or if the execution by other parties is defective, void or voidable.

13 Amendment

This document may only be varied or replaced by a document executed by the parties.

14 Governing law and jurisdiction

- (a) This document and the transactions contemplated by this document are governed by and are to be construed in accordance with the laws applicable in New South Wales.
- (b) Each party irrevocably and unconditionally submits to the non-exclusive jurisdiction of the courts exercising jurisdiction in New South Wales and any courts which have jurisdiction to hear appeals from any of those courts and waives any right to object to any proceedings being brought in those courts.

15 Counterparts

This document may consist of a number of counterparts and the counterparts taken together constitute one and the same instrument.

16 General

16.1 Construction

Unless expressed to the contrary, in this document:

- (a) words in the singular include the plural and vice versa;
- (b) any gender includes the other genders;
- (c) if a word or phrase is defined its other grammatical forms have corresponding meanings;
- (d) 'includes' means includes without limitation;
- (e) no rule of construction will apply to a clause to the disadvantage of a party merely because that party put forward the clause or would otherwise benefit from it;
- (f) a reference to:
 - a person includes a partnership, joint venture, unincorporated association, corporation and a government or statutory body or authority;
 - (ii) a person includes the person's legal personal representatives, successors, assigns and persons substituted by novation;
 - (iii) any legislation includes subordinate legislation under it and includes that legislation and subordinate legislation as modified or replaced;
 - (iv) an obligation includes a warranty or representation and a reference to a failure to comply with an obligation includes a breach of warranty or representation;
- (g) Capitalised terms not defined in this document have the meanings given to them in the Planning Agreement.

16.2 Headings

Headings do not affect the interpretation of this document.

16.3 Effect of execution

This document is not binding on any party unless it or a counterpart has been duly executed by each party.

Execution

| Executed as a deed/agreement. | |
|---|-----------------------|
| Minister | |
| Signed, sealed and delivered by The Minister for Planning in the presence of:) | |
| Signature of Witness | Signature of Minister |
| Name of Witness | Name of Minister |
| Developer Signed sealed and delivered for and on behalf of Goodman Property Services (Aust) Pty Limited by its attorney under power of attorney BookNowho states | |
| that no notice of revocation of the power of attorney has been received in the presence of: | |
| Signature of witness | |
| Name of witness (BLOCK LETTERS) | |
| Address of witness | |

First Landowner

| Signed sealed and delivered for and on behalf of The Trust Company Limited as custodian for the Carter Street Trust by its attorney under power of attorney BookNowho states that no notice of revocation of the power of attorney has been received in the presence of: | |
|--|--------------------------|
| Signature of witness | Attorney |
| Name of witness (BLOCK LETTERS) | Name of Attorney (print) |
| Address of witness | |
| Second Landowner | |
| Signed sealed and delivered for and on behalf of The Trust Company Limited as custodian for the Hill Road Trust by its attorney under power of attorney BookNowho states that no notice of revocation of the power of attorney has been received in the presence of: | |
| Signature of witness | Attorney |
| - | • |
| Name of witness (BLOCK LETTERS) | Name of Attorney (print) |
| Address of witness | |

Third Landowner

Signed sealed and delivered for and on behalf of The Trust Company Limited as custodian for the IBC Trust by its attorney under power of attorney Book.....No...who states that no notice of revocation of the power of attorney has been received in the presence of: Signature of witness Attorney Name of witness (BLOCK LETTERS) Name of Attorney (print) Attorney Address of witness **Trustees** Signed sealed and delivered for and on behalf of Tallina Pty Ltd as trustee for the Carter Street Trust by its attorney under power of attorney Book.....No...who states that no notice of revocation of the power of attorney has been received in the presence of: Signature of witness Attorney Name of witness (BLOCK LETTERS) Name of Attorney (print)

Address of witness

| behalf of Tallina Pty Ltd as trustee for the Hill Road Trust by its attorney under power of attorney BookNowho states that no notice of revocation of the power of attorney has been received in the presence of: | |
|---|--------------------------------------|
| Signature of witness | Attorney |
| | , |
| Name of witness (BLOCK LETTERS) | Name of Attorney (print) |
| Address of witness | |
| Signed sealed and delivered for and on behalf of Tallina Pty Ltd as trustee for the IBC Trust by its attorney under power of attorney BookNowho states that no notice of revocation of the power of attorney has been received in the presence of: | |
| Signature of witness | Attorney |
| Name of witness (BLOCK LETTERS) | Name of Attorney (print) Attorney |
| Address of witness | |
| New Party | |
| Executed by [Insert company name) and ABN]:) | |
| Company Secretary/Director | Director |
| Name of Company Secretary/Director (print) | Name of Director (print) |

Annexure B

Plan





Annexure C

Education Land

[Survey to be provided prior to execution of this document]



Annexure D

Form of Bank Guarantee



FORM OF BANK GUARANTEE

Date: [Insert Date]

TO: [Note: Insert Name and Address of Beneficiary]

Dear Sirs

Bank Guarantee

NAME OF LENDER (the "Lender") has pleasure in detailing the particulars of our Bank Guarantee issued in your favour.

- 1. At the request of [*] (the "Account Party") and in consideration of [*] (the "Beneficiary") accepting this Bank Guarantee in relation to [Note: Insert Details of transaction supported by Bank Guarantee] (the "Secured Transaction"), the Lender unconditionally undertakes to pay any sum or sums which may from time to time be demanded under this Bank Guarantee by the Beneficiary to a maximum total sum of A\$[*].
- 2. This Bank Guarantee is to continue until the first to occur of:
 - (a) its return to the Lender; or
 - (b) payment at any time to the Beneficiary by the Lender of the maximum amount available under this Bank Guarantee.
- 3. If the Lender receives a written demand at its office at [ADDRESS OF LENDER] no later than 5pm on the Expiry Date and the demand:
 - (a) purports to be signed on behalf of the Beneficiary;
 - (b) demands payment of all or any part of that sum; and
 - (c) is accompanied by a declaration complying with Clause 4,

the Lender will pay to the Beneficiary within [*] Business Days of demand, the amount demanded without further reference to the Account Party and despite any notice to the contrary from the Account Party.

- 4. A declaration accompanying a demand under this Bank Guarantee must state that:
 - (a) the declarants are 2 officers of the Beneficiary, making the declaration on behalf of the Beneficiary;
 - (b) the declarants have authority to make the declaration on behalf of the Beneficiary;
 - (c) the declaration is made under the Bank Guarantee dated [*];
 - (d) the amount claimed is not more than the maximum amount available under this Bank Guarantee;

- (e) the amount claimed represents an amount or amounts remaining unpaid to the Beneficiary in respect of the Secured Transaction; [and
- (f) demand for payment of that amount has been made by the Beneficiary on the Account Party and the demand remains unsatisfied].
- 5. This Bank Guarantee is governed by the laws of New South Wales.
- 6. This Bank Guarantee may not be assigned without prior written consent of the Lender.

| Executed in [*]. On behalf of NAME OF LENDER: | | | |
|---|--------------------|--|--|
| Authorised Officer | Authorised Officer | | |
| Print Name | Print Name | | |

Annexure E

Form of Caveat



Form: 08X 1309 Edition:

Licence: 04-08-443

Licensee: Corrs Chambers Westgarth

Leave this space clear. Affix additional pages to the top left-hand corner.

Prohibiting Recording of a Dealing or Plan or Granting of a Possessory Application **New South Wales** Section 74F Real Property Act 1900

PRIVACY NOTE: Section 31B of the Real Property Act 1900 (RP Act) authorises the Registrar General to collect the information required by this form for the establishment and maintenance of the Real Property Act Register. Section 96B RP Act requires that the Register is made available to any person for search upon payment of a fee, if any.

| STA | AMP DUTY | Office of State Revenue use only | | |
|-----|---|---|--|--|
| (A) | TORRENS TITLE | If the claim relates to less than the whole of the land in the Torrens Title, a description of the part or premises affected, consistent with the claim set out on page 2 of this form and in the terms specified by Schedule 4 of the current Real Property Regulation, is required. Folio identifiers 16/225350, 17/225350, 18/225350, 19/225350, 23/225350, 24/225350, 25/225350, 26/225350, 27/225350, 28/225350, 2/234801 and 301/541070 | | |
| (B) | REGISTERED DEALING | Number Torrens Title | | |
| (C) | LODGED BY | Document Collection Box Reference (optional): | | |
| (D) | REGISTERED PROPRIETOR | Show only the registered proprietor(s) against whom the claim is made: insert the full name and address The Trust Company Limited C/- Goodman Property Services (Aust) Pty Limited Level 17, 60 Castlereagh Street Sydney NSW Postcode: 2000 | | |
| (E) | CAVEATOR | Insert the full name and address (residential if individual/registered office if body corporate) The Minister for Planning Department of Planning and Environment 23-33 Bridge Street Sydney NSW Postcode: 2000 | | |
| (F) | ADDRESS IN NEW SOUTH WALES FOR SERVICE OF NOTICES ON THE CAVEATOR | IMPORTANT NOTE: The address <i>must</i> be a street address. If desired, a Document Exchange box in NSW may be provided <i>in addition</i> . If the caveator's name or address for service of notices changes, notification <i>must</i> be lodged on form 08CX. Name: Louise Camenzuli Street Address: Corrs Chambers Westgarth Level 17, 8 Chifley 8-12 Chifley Square Sydney NSW NSW postcode: 2000 Document Exchange Box in NSW (additional): | | |
| (G) | ACTION PROHIBITED | List by number only the items in Schedule 2 prohibited by this caveat 1, 2, 5 and 7 | | |

(H) The caveator claims to be entitled to the estate or interest specified in Schedule 1 in the above land/registered dealing by virtue of the instrument/facts set out in that schedule and prohibits the Registrar General from taking, with respect to the land/registered dealing, the action specified above unless the caveator has consented in writing or this caveat has lapsed or been withdrawn.

WARNING: care should be exercised in completing a caveat form. An insupportable caveat may be challenged in the Supreme Court; compensation may be awarded for lodging a caveat without justification (section 74P Real Property Act 1900). Failure to observe the requirements of regulations 7 and 8 of the current Real Property Regulation may make the caveat invalid.

| (I) | SCHEDIII F 1 | Estate or interest | claimed |
|-----|--------------|--------------------|---------|
| | | | |

| Particulars of the estate or interest in the land/registered dealing | | |
|--|---|---|
| Legal interest | | |
| By virtue of the instrument referred to be | elow | |
| Nature of Instrument | Date | Parties |
| Planning agreement | [] | The Minister for Planning Goodman Property Services (Aust) Pty Limited The Trust Company Limited Tallina Pty Ltd |
| By virtue of the facts stated below | - | |
| By virtue of the charge over the whole of agreement | of the land granted by The T | rust Company Limited to the Minister for Planning in the above planning |
| Schedule 1. The registration or recording of any print in Schedule 1. The registration of delimitation plan¹ The granting of any possessory application. The recording in the register of any control of the granting of an application to exting the recording of the granting of an application to exting the register. | y dealing other than a plan plan other than a delimitation No. cation ² with respect to the late lealing affecting the estate of anguish the restrictive covens | and in the Torrens Title referred to above. In interest of which the caveator is registered proprietor. In the torrent of which the caveator is registered proprietor. |
| | ation and belief slid claim to the estate or inte the address of the registere | erest set out in Schedule 1 d proprietor is the correct address. he endorsed consent of the registered proprietor/possessory applicant. |

(K)

(J)

I make this solemn declaration conscientiously believing the same to be true and by virtue of the Oaths Act 1900 and I certify this caveat to be correct for the nurnoses of the Real Property Act 1900

| COLLECT | ior the purposes of the real inoperty Ac | 1300. | |
|---|---|----------|--|
| Made a | and subscribed at Sydney in the State of N | ISW on | in the presence of- |
| ☐ Jus | stice of the Peace (J.P. Number: |) | ☐ Practising Solicitor |
| Oth | ner qualified witness Solicitor, | | |
| ** who | certifies the following matters concerning | the ma | king of this statutory declaration by the person who made it: |
| 1. | I saw the face of the person / I did not se | e the fa | ace of the person because the person was wearing a face covering, but I am satisfied |
| | that the person had a special justification | for no | t removing the covering; and |
| 2. I have known the person for at least 12 months / I have confirmed the person's identity using an identification document and the | | | |
| | document I relied on was: | | |
| Signature of witness: | | | Signature of declarant: |
| | | | Capacity of declarant if other than the caveator: |

- A plan defining the boundaries of land in a limited folio of the Register. See Part IVB Real Property Act 1900.
- 2 An application made by a person claiming title to land by virtue of adverse possession. See Part VIA Real Property Act 1900
- 3 As the services of qualified witness cannot be provided at lodgment, the declaration should be signed and witnessed prior to lodgment. **If made outside NSW, cross out the witness certification. If made in NSW, cross out the text which does not apply.
- Only one capacity can be selected, either registered proprietor or possessory applicant, cross out whichever does not apply.